ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

IN THE MATTER OF THE *COMPANIES' CREDITORS* ARRANGEMENT ACT, RSC 1985, c C-36, AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF SHAW-ALMEX INDUSTRIES LIMITED AND SHAW ALMEX FUSION, LLC

Applicants

FACTUM OF THE APPLICANTS (Re: Motion Approval And Vesting Order – returnable July 18, 2025)

July 17, 2025

RECONSTRUCT LLP

80 Richmond Street West, Suite 1700 Toronto, ON M5H 2A4

Brendan Bissell LSO No. 40354V

bbissell@reconllp.com Tel: 416.613.0066

Simran Joshi LSO No. 89775A

sjoshi@reconllp.com Tel: 416.613.6589

Fax: 416.613.8290

Lawyers for the Applicants

TO: THE SERVICE LIST

TABLE OF CONTENTS

PART I	- OVERVIEW	1
PART II	- FACTS	3
A.	Background of the Restructuring Proceeding	3
B.	The SISP	4
C.	The Transaction	6
PART II	I – ISSUES	7
PART I	/ – LAW & ARGUMENT	В
A.	The Court Should Approve the Asset Purchase Agreement and the Transaction	3
B.	The Court Should Approve the Extended Stay Period10	6
C.	The Court Should Approve the Third Amendment and Increase the DIP Lender's Charge	е
	18	
D.	The Court Should Approve the Sealing Provision1	9
E.	The Court Should Approve the Monitor's Activities in the Reports and the Professional	al
Fees	of the Monitor20	C
PART V	- RELIEF REQUESTED2	2
Schedu	le "A"2	3
Schedu	le "B"2	5

PART I – OVERVIEW

- 1. On May 13, 2025, the Ontario Superior Court of Justice (Commercial List) (the "Court") granted an initial order (the "Initial Order") pursuant to the *Companies' Creditors Arrangement Act*, R.S.C. 1985, c. C-36, as amended (the "CCAA") in favour of the Shaw-Almex Industries Limited ("SAIL") and Shaw Almex Fusion, LLC ("Fusion" and together with SAIL, the "Applicants"). The Initial Order, among other things, granted the Applicants protection, imposed a stay of proceedings up to and including May 30, 2025 (the "Stay of Proceedings"), appointed FTI Consulting Canada Inc. ("FTI") as monitor of the Applicants with enhanced powers (in such capacity, the "Monitor"), and authorized the Applicants to borrow up to a maximum principal amount of \$1,836,000 under a facility (the "DIP Facility") pursuant to an amended and restated interim financing term sheet (the "Amended DIP Term Sheet").
- 2. On the same date, the Court granted an order (the "SISP Approval Order") approving the sale and investment solicitation process (the "SISP").
- 3. By orders dated May 30 and June 27, 2025, the Court granted two amendments to the DIP Facility to enable increased borrowing, which is now capped at a maximum principal amount of \$3,646,500. The Court also extended the Stay of Proceedings to August 1, 2025.
- 4. The deadline for bidders to submit a binding offer in the SISP was June 12, 2025. After reviewing the "Qualified Purchase Bids", the Monitor after consulting with the DIP Lender determined that the bid submitted by an affiliate of Rema Tip Top of America, Inc. was the "Successful Bid" (as such terms are defined in the SISP Approval Order).
- 5. The Applicants now seek the following orders:
 - (a) an approval and vesting order (the "AVO") substantially in the form of the draft order included at tab 3 of the Applicants' Motion Record that, among other things:
 - (i) approves the Asset Purchase Agreement dated as of July 10, 2025 (the "Asset Purchase Agreement"), between the Applicants, as vendors, and, Almex Canada, Limited (the "Purchaser"), as purchaser;
 - (ii) approves the transaction contemplated by the Asset Purchase Agreement (the "**Transaction**") and authorizes the Applicants to take such additional

- steps and execute such additional documents as necessary or desirable for the completion of the Transaction;
- (iii) in respect of the Transaction, seeks a declaration that neither Mr. Timothy Shaw nor Mrs. Pamela Shaw, nor any person acting on their behalf or in which they have a legal, economic or beneficial interest, has any legal, economic or beneficial interest in or to the Intellectual Property (as defined in the Asset Purchase Agreement) listed on Schedule 2.1(i) of the Asset Purchase Agreement;
- (iv) seals Confidential Exhibit "1" (the "Confidential Exhibit") to the affidavit of Andrew Hustrulid sworn July 14, 2025 (the "Fourth Hustrulid Affidavit"), which contains an unredacted copy of the Asset Purchase Agreement, until or further Order of the Court; and
- (v) seals the confidential supplement (the "Confidential Supplement") to the Third Report of the Monitor, to be filed (the "Third Report") until further order of the Court; and
- (b) an order substantially in the form attached as tab 5 (the "**Ancillary Order**") to the Applicants' Motion Record that, among other things:
 - (i) extends the Stay of Proceedings up to and including September 5, 2025 (the "Extended Stay Period");
 - (ii) authorizes and empowers the Applicants (or the Monitor on behalf of the Applicants) to enter into the Third Amendment to the Amended and Restated DIP Facility Loan Agreement made between the Applicants and RBC dated as of July 17, 2025 (the "Third Amendment") and borrow up to a maximum principal amount of \$4,641,000 under an amended DIP Facility (the "Third Amended DIP Facility") from the Royal Bank of Canada (in its capacity as lender, the "DIP Lender") to finance the Applicants' working capital requirements and to pay the costs and expenses of this proceeding;
 - (iii) increases the quantum of the DIP Lender's Charge to the maximum amount of \$4,641,000, plus fees and interest;

- (iv) approves the First Report of the Monitor dated May 27, 2025 (the "First Report"), the Supplement to the First Report of the Monitor dated May 28, 2025 (the "Supplement"), the Second Report of the Monitor dated June 27, 2025 (the "Second Report") the Third Report of the Monitor dated July 16, 2025 (the "Third Report"), and the Supplement to the Third Report of the Monitor dated July 17, 2025 and (collectively with the First Report, Supplement, Second Report, and Third Report, the "Reports") and the activities of the Monitor and its counsel in the Reports; and
- (v) approves the fees and disbursements of the Monitor and its legal counsel (the "Professional Fees"), as described in the fee affidavits appended to the Third Report.
- 6. The Applicants submit that it is appropriate to grant the requested relief including the approval of the Asset Purchase Agreement. The SISP thoroughly canvassed the market for a sale and/or investment of the Applicants and their assets. The Asset Purchase Agreement represents the best available outcome that maximizes value and permits the business of the Applicants to continue as a going-concern for the benefit of their employees, suppliers and customers.
- 7. The Monitor supports the within motion and the Applicants' request for the AVO and Ancillary Order.

PART II - FACTS

8. Capitalized terms not otherwise defined herein have the meaning ascribed to them in the Fourth Hustrulid Affidavit and the Asset Purchase Agreement.

A. Background of the Restructuring Proceeding

9. The Applicants are in the business of providing customized solutions for all aspects of conveyor belt systems (the "Business").¹

¹ Affidavit of Andrew Hustrulid sworn July 14, 2025, Tab 2, Motion Record of the Applicants dated July 14, 2025 ("**MR**") (the "**Fourth Hustrulid Affidavit**") at <u>para. 4</u>.

4

10. SAIL is the parent company of a global enterprise operating under the "Shaw Almex" name (the "**Almex Group**"). The Almex Group has 15 locations, plus exclusive distributors, across six continents worldwide. Through these locations, the Applicants service customers across 123 countries worldwide.²

11. Since the granting of the Initial Order, the Applicants have undertaken and implemented a comprehensive operational restructuring initiative aimed at stabilizing and improving the financial performance of their Business. This restructuring has included a thorough review and streamlining of the Applicants' operations, reduction of workforce, optimization of workforce allocation, and the sale of a majority of Fusion's inventory and equipment in accordance with the authority granted by this Court under paragraph 12 of the Initial Order.³ As a direct result of these efficiencies, the Applicants have been able to increase production capacity and throughput at their principal global manufacturing facility located at 17 Shaw Almex Road, Parry Sound, Ontario.⁴

B. The SISP

Overview of the SISP

12. As noted above, on May 13, 2025, the Court granted the SISP Approval Order. The SISP was developed in consultation with and overseen by the Monitor. The SISP built on efforts already taken by SAIL to informally canvas the market more than a year prior to the formal commencement of the SISP.⁵ The SISP was intended to complement the Applicants' ongoing operational restructuring efforts and to maximize value for the Applicants' stakeholders by widely exposing the Applicants' Business and Property to the market. ⁶

- 13. The SISP was administered by the Monitor with the assistance of the Applicants and in consultation with the DIP Lender, in accordance with the terms of the SISP Approval Order. The SISP was designed as a two-phase process occurring over a period of up to six weeks:
 - (a) phase 1 called for non-binding letters of interest by no later than 5:00 pm ET on May 22, 2025;⁷ and

² Fourth Hustrulid Affidavit at para. 5.

³ Fourth Hustrulid Affidavit, Exhibit "B", Initial Order, para. 12.

⁴ Fourth Hustrulid Affidavit at paras. 12-13.

⁵ Third Report of the Monitor dated July 16, 2025 ("Third Report") at para. 28.

⁶ Fourth Hustrulid Affidavit at paras. 14-15.

⁷ Fourth Hustrulid Affidavit at para. 18; Third Report at para. 24.

(b) phase 2 invited "Qualified Bidders" to submit binding offers by no later than 5:00 pm ET on June 12, 2025.8

Implementation of the SISP

- 14. The Monitor's extensive solicitation efforts are described in the Fourth Hustrulid Affidavit and the Third Report delivered in support of this motion. In particular:
 - (a) the proposal trustee prepared a list of approximately 147 known prospective purchasers and investors ("Prospective Bidders"), which formed the basis of the proposal trustee (and later Monitor's) solicitation efforts;
 - (b) all Prospective Bidders were sent a solicitation letter summarizing the acquisition and investment opportunity with respect to the Business and Property and a form of non-disclosure agreement ("NDA"); and
 - approximately 18 Prospective Bidders entered into NDAs and were granted access (c) to a virtual data room containing due diligence information and documentation in relation to the Applicants.9
- 15. The Monitor broadly canvassed the market in the course of the SISP, and it worked diligently with prospective purchasers to respond to all questions and inquiries received in respect of the Almex Group to ensure that prospective purchasers had the information necessary to consider a transaction.¹⁰
- 16. Upon review of the Qualified Purchase Bids and Qualified Investment Bids, the Monitor, in consultation with the DIP Lender, determined that accepting the Qualified Purchase Bid submitted by an affiliate of Rema Tip Top America, Inc. was in the best interests of the Applicants and their stakeholders.11

⁸ Fourth Hustrulid Affidavit at para. 20 and 23; Third Report at para. 24.

⁹ Fourth Hustrulid Affidavit at <u>para. 24</u>.
¹⁰ Fourth Hustrulid Affidavit at <u>para. 26</u>; Third Report at <u>para. 27</u>.

¹¹ Fourth Hustrulid Affidavit at para. 28.

C. The Transaction

17. The Purchaser is seeking to purchase substantially all of the Applicants' assets under the Asset Purchase Agreement. The particulars of the Asset Purchase Agreement are set out at paragraph 30 of the Fourth Hustrulid Affidavit.¹²

18. The Purchased Assets include:

- (a) the Applicants' Intellectual Property. The Applicants own the Intellectual Property and have patent registrations with the relevant intellectual property offices. However, many of the Applicants' trademarks and tradenames are not registered. The Purchaser places significant value on the Applicants' Intellectual Property and the purchase price that the Purchaser is prepared to pay is heavily dependent on the Purchaser's ability to acquire the Applicants' Intellectual Property free of any competing claims;¹³ and
- (b) all of the Equity Interests of SAIL in the capital of the Purchased Subsidiaries, being Fonmar Group, S.L.; Shaw Almex Mine Equip. (Tianjin) Co. Ltd.; Shaw Almex Pacific Pty Ltd.; Shaw Almex Europe B.V.; Shaw Almex Chile SpA; Almex Peru S.A.C.; and PT. Shaw Almex Indonesia.
- 19. Should the Transaction be approved, the Business will continue as a going concern, enabling the continued employment of certain employees of the Applicants and the ongoing provision of services to customers, and the purchase price will generate significant proceeds for the benefit of the Company's secured creditors.

D. Approval of the Monitor's Professional Fees and Reports

20. The Monitor is seeking approval of the Professional Fees of FTI, , in its capacity as Monitor, incurred during the period May 1, 2025 through to and including June 30, 2025, 14 and

¹² Fourth Hustrulid Affidavit at para. 30.

¹³ Third Report at paras. 47 and 48.

¹⁴ Third Report, Appendix "A", Fee Affidavit of Jeffrey Rosenberg affirmed July 16, 2025.

approval of the fees and disbursements of its counsel, Stikeman Elliott LLP ("**Stikeman**"), incurred during the period of May 5, 2025 through to and including July 11, 2025.¹⁵

- 21. In support of this motion, the Monitor delivered its Third Report, which attaches affidavits from representatives of the Monitor and its legal counsel that provide a comprehensive listing of the accounts sought to be passed and summary tables identifying the individual professionals who have worked on this matter, their hourly billing rates and total numbers of hours worked, among other information.¹⁶
- 22. The fees and disbursements of FTI from May 1, 2025 through to and including June 30, 2025, total \$575,306.00 (exclusive of HST).
- 23. The fees and disbursements Stikeman from May 5, 2025 through to and including July 11, 2025, total \$609,936.45 (exclusive of HST).

PART III - ISSUES

- 24. The issues to be determined are whether the Court should:
 - approve the Asset Purchase Agreement and the Transaction contemplated therein pursuant to the terms of the proposed AVO;
 - (b) extend the Stay of Proceedings up to and including September 5, 2025;
 - (c) approve the Third Amendment to the DIP Facility and the increase in DIP Lender's Charge;
 - (d) approve the sealing provision; and
 - (e) approve the Monitor's Reports and Professional Fees.

¹⁵ Third Report, <u>Appendix "B"</u>, Fee Affidavit of Maria Konyukhova affirmed July 16, 2025. There is one docket entry included in Stikeman Elliott's invoices that is from May 3, 2025. This docket entry was not approved as part of Stikeman Elliott's prior fee approval, which approved Stikeman Elliott's fees and disbursements up to and including May 4, 2025. ¹⁶ <u>Ibid</u>.

PART IV - LAW & ARGUMENT

A. The Court Should Approve the Asset Purchase Agreement and the Transaction

- 25. It is well-established that this Court has jurisdiction to approve a sale of all or substantially all of the assets of a debtor company in a CCAA proceeding where such sale is in the best interests of stakeholders generally. The sale of a business as a going concern during a CCAA proceeding is consistent with the purposes of the CCAA.¹⁷
- 26. Pursuant to Section 36 of the CCAA, the Court has the jurisdiction to approve a sale transaction within the context of CCAA proceedings. Subsection 36(3) sets out the following factors for the Court to consider when determining whether to authorize a sale of assets by a debtor company in a CCAA proceeding. The criteria is non exhaustive and the Court must look at the proposed transaction as a whole and decide whether it is appropriate, fair and reasonable:
 - (a) whether the process leading to the proposed sale or disposition was reasonable in the circumstances;
 - (b) whether the monitor approved the process leading to the proposed sale or disposition;
 - (c) whether the monitor filed with the court a report stating that in their opinion the sale or disposition would be more beneficial to the creditors than a sale or disposition under a bankruptcy;
 - (d) the extent to which the creditors were consulted;
 - (e) the effects of the proposed sale or disposition on the creditors and other interested parties; and
 - (f) whether the consideration to be received for the assets is reasonable and fair, taking into account their market value.¹⁸

¹⁷ Re Nortel Networks Corp. (2009), 2009 CanLII 39492 (Ont. Sup. Ct. J.) at <u>paras. 35-40</u> and <u>48</u> [Nortel"]; Re Brainhunter Inc. (Re) CanLII 67659 (ON SC) at <u>para. 12</u>; Re Canwest Publishing Inc./Publications Canwest Inc., 2010 ONSC 2870 at para. 13. [Canwest]

¹⁸ Companies' Creditors Arrangement Act, R.S.C. 1985, c. C-36, as amended ("CCAA"), <u>s. 36(3)</u>; Re Nelson Education Limited, 2015 ONSC 5557 at <u>para. 38</u> [Nelson]; Re Bloom Lake, 2015 QCCS 1920 <u>at paras. 25-26</u>; Target Canada Co. (Re), 2015 ONSC 1487 [Target] at <u>para. 16</u>.

- 27. In *Canwest*, Justice Pepall held that the criteria enumerated in subsection 36(3) of the CCAA largely overlapped with the traditional common law criteria established in *Royal Bank of Canada v Soundair Corp* for approval of a sale of assets in an insolvency scenario and remain relevant when considering the statutory test, as follows (collectively, the "*Soundair Principles*"):
 - (a) whether the Court-appointed officer has made sufficient effort to get the best price and has not acted improvidently;
 - (b) the interest of all parties;
 - (c) the efficacy and integrity of the process by which the offers are obtained; and
 - (d) whether there has been unfairness in the working out of the process.¹⁹
- 28. The above criteria in subsection 36(3) and the *Soundair* Principles support this Court's approval of the Transaction for the following reasons:
 - (a) The SISP was developed and implemented by the Monitor. The SISP was developed with significant input from the Monitor and was administered by the Court-appointed Monitor.
 - (b) The SISP was fair and transparent. The Monitor administered the SISP in accordance with the terms of the SISP Approval Order. All potential bidders were treated in an even-handed and fair manner. In particular, all potential bidders, including the Purchaser, were provided with the same information and ability to conduct due diligence.
 - diligent and comprehensive efforts to obtain the highest and best price available in the circumstances. The process was designed to canvass a wide range of potential transaction structures, including a sale of, or investment in, the Business. To ensure broad market exposure, the Monitor prepared a detailed solicitation letter outlining the acquisition opportunity in respect of the Business and the Property, which was distributed to approximately 147 known and credible prospective purchasers and investors. The SISP's duration (42 days from the

¹⁹ Canwest, supra at <u>para. 13</u>; Royal Bank of Canada v Soundair Corp., 1991 ONCA 2727 [Soundair] at <u>para. 16</u>. See also Target at <u>paras. 14-17</u>.

launch of the SISP to the end of Phase 2) was adequate and typical for an insolvency proceeding of this nature, particularly given that the SISP was not a *de novo* process in that it built on SAIL's approximately year-old effort to canvas the market for strategic alternatives.²⁰

- (d) The Transaction is in the best interests of stakeholders. The purchase price as contemplated in the Asset Purchase Agreement provides maximum value for the benefit of the Applicants' stakeholders. It is superior to the other bids that were submitted in the course of the SISP and represents the market price for the Applicants' business in the circumstances. The Transaction provides material value for the Applicants' creditors and permits the Business to continue under a new entity that will preserve employment for most of the Applicants' employees, provide continued services to its customers, provide for the continuation of the Assumed Contracts, and maintain ongoing revenue for the Applicants' critical vendors. The DIP Lender consented to the Applicants entering into the Asset Purchase Agreement. Page 1972.
- (e) The Monitor supports the Transaction. In *Nelson*, the Court indicated that the monitor's "blessing" of a sale process is an important factor to consider.²³ The Monitor believes the Transaction presents the best possible outcome for the stakeholders in the circumstances. The Monitor believes that a bankruptcy and/or liquidation would be a suboptimal outcome as compared to the Transaction because the Applicants' secured creditors will likely recover significantly less proceeds in a liquidation, employees would be terminated, and customers and vendors would suffer losses.
- (f) There is no prejudice to any creditor as a consequence of the Transaction.
 As of the date hereof, the Applicants are not aware of any objection to the Transaction.
- (g) There is no viable alternative to the Transaction. Based on the results of the SISP, it appears unlikely that a better transaction would emerge to acquire the

²⁰ Third Report at para. 40.

²¹ Third Report at para. 41.

²² <u>Ibid</u>

²³ Nelson, supra at para. 38.

Business as a going concern should the Transaction not proceed. As such, the Transaction represents the best option to avoid the Applicants ceasing operations and liquidating their assets in a bankruptcy. In a bankruptcy, the Business would likely be forced to shut down, which would result in irreparable destruction of asset value given that the Applicants' primary assets – its Intellectual Property, goodwill, and equipment – require an operating Business to retain their value. In such circumstances, employees lose their employment, customers lose their continued services, and the ongoing arrangements with vendors would not be preserved.

- (h) The consideration given by the Purchaser is fair and reasonable in the circumstances. The purchase price is fair and reasonable in the circumstance. The Transaction is a value-maximizing transaction that monetizes substantially all of the assets for the benefit of the Applicants' creditors.
- 29. Accordingly, the Applicants respectfully submit that this Court should approve the Transaction.

Intellectual Property

- 30. As part of the AVO, the Applicants seek a declaration that neither Timothy Shaw nor Pamela Shaw, nor any person acting on their behalf or in which they have any legal, economic or beneficial interest, has any legal, economic or beneficial interest in or to the Intellectual Property listed on Schedule 2.1(i) to the Asset Purchase Agreement (the "Intellectual Property Schedule"). This relief is being sought at the request of the Purchaser and as a condition to the completion of the Transaction.
- 31. The Applicants' Business operations are fundamentally reliant on the use of their Intellectual Property, which include various registered and unregistered wordmarks, trademarks, and other proprietary assets under which the Applicants conduct their commercial activities. It is essential to the Purchaser that it obtain unambiguous, exclusive, and unencumbered rights to the Applicants' Intellectual Property.
- 32. In *S.A. v. Metro Vancouver Housing Corp.*, the Supreme Court of Canada summarized the legal test for the granting of declaratory relief: "Declaratory relief is granted by the courts on a discretionary basis, and may be appropriate where (a) the court has jurisdiction to hear the issue, (b) the dispute is real and not theoretical, (c) the party raising the issue has a genuine interest in

its resolution, and (d) the responding party has an interest in opposing the declaration being sought."²⁴

- 33. The Applicants submit that the legal test for the granting of the declaration with respect to the Applicants' Intellectual Property is satisfied in the circumstances of this case:
 - (a) The Court has the jurisdiction under section 11 of the CCAA, on notice to any other person or without notice as it may see fit, to make any order that it considers appropriate in the circumstances.²⁵ Further, a judge of the Superior Court of Justice is authorized under section 97 of the *Courts of Justice Act* to make binding declarations of right, whether or not any consequential relief is or could be claimed²⁶. This Court will grant declaratory relief in the context of a CCAA proceeding where the motion materials provide the necessary evidentiary foundation.²⁷
 - The need for declaratory relief in relation to the Applicants' Intellectual Property is (b) a real and important issue given concerns expressed by FTI, in its capacity as proposal trustee and subsequently as Monitor, that Mr. Shaw may be currently using, or intends to use, the Applicants' Intellectual Property in connection with a competing business. As detailed in its prior Reports to Court, the Monitor has been advised that Mr. Shaw transferred certain of SAIL's property to other parties outside of the ordinary course of business, that Mr. Shaw requested and received portable hard drives containing copies of SAIL's Intellectual Property, and that Mr. Shaw intended to establish a new business to compete with the Applicants using the Applicants' own Intellectual Property.²⁸ Such conduct could constitute a direct infringement on the rights that the Applicants intend to transfer to the Purchaser under the Asset Purchase Agreement. These concerns, and others, with respect to actions by Mr. Shaw that misappropriate assets from the Applicants and harm the Applicants' restructuring efforts led the Monitor to request, and the Court to grant, the Property Preservation Order. Furthermore, the Monitor has a pending

²⁴ S.A. v. Metro Vancouver Housing Corp., 2019 SCC 4 (CanLII), [2019] 1 SCR 99 at para. 60.

²⁵ CCAA, supra <u>s. 11</u>.

²⁶ Courts of Justice Act, RSO 1990, c C.43, s. 97.

²⁷ Nortel Networks v. Ernst & Young Inc., 2024 ONSC 6386 2024 CarswellOnt 18306 at paras, 6 and 7.

²⁸ Endorsement of Justice Dietrich dated May 30, 2025 at <u>paras. 20</u> and <u>27;</u> First Report of the Monitor dated May 27, 2025 at <u>para. 51</u>.

motion before the Court seeking a determination that Mr. Shaw is in contempt of the Property Preservation Order.

- (c) The Applicants, the Monitor and the Purchaser have a genuine interest in confirming the Applicants' ownership of the Intellectual Property to the exclusion of Mr. and Mrs. Shaw. The value that the Purchaser is prepared to pay for the Applicants' Business and assets are heavily dependent on the Applicants' Intellectual Property. To safeguard the integrity of the Transaction and ensure the Purchaser receives the full value of the Applicants' Business, it is essential that the Purchaser obtain unambiguous, exclusive, and unencumbered rights to the Applicants' Intellectual Property. The potential for any post-closing claims by Mr. or Mrs. Shaw poses a material risk that could undermine the Purchaser's ability to capitalize on the assets it is acquiring. The granting of the declaration is necessary for, and will facilitate the completion of the Transaction, which will maximize the value of the Applicants' Business and assets and result in a going concern outcome for the benefit of a broad range of stakeholders.
- (d) The Applicants believe that all of the Intellectual Property listed on the Intellectual Property Schedule is owned by SAIL or its Chinese affiliate, Shaw Almex Mine Equip. (Tianjin) Co. Ltd. (which is a Purchased Subsidiary under the Transactions). ²⁹ The Applicants have consistently used the Intellectual Property in their operations and maintained secrecy and contractual controls over the Intellectual Property. ³⁰ The Applicants do not believe, and there is no evidence, that Mr. Shaw or Mrs. Shaw have any legal, economic or beneficial interest in the Intellectual Property listed on the Intellectual Property Schedule. If Mr. Shaw or Mrs. Shaw have a different view, they will have a clear interest in opposing the declaration being sought. Mr. Shaw (through counsel) and Mrs. Shaw were served with the Applicants' motion record, and the Applicants' counsel contacted Mr. Shaw's counsel and Mrs. Shaw to inform them specifically that the declaration was being sought as part of the proposed Approval and Vesting Order.
- 34. As described by the Alberta Court of King's Bench in *Taber Water*, vesting orders provide purchasers with certainty of title which de-risks assets and encourages purchasers to pay

²⁹ Fourth Hustrulid Affidavit at para. 41.

³⁰ Third Report at para. 47.

increased prices. Vesting orders are essential to maximizing the return of creditors of an insolvent entity. Competing claims to ownership of assets in the possession of a debtor should be identified and resolved prior to their sale in an insolvency proceeding so that the assets may be sold without a cloud over title.³¹

- 35. It is important that the proponent of a restructuring transaction and all stakeholders have a clear understanding of what assets a business will own upon completion of the restructuring transaction and that claims that are not advanced in a timely manner are barred and extinguished. Allowing a claimant, who was aware that a debtor in an insolvency proceeding was purporting to sell an asset which the claimant believed it owned, to lay claim to that asset after a restructuring is approved and implemented would be an absurd result.³²
- 36. While the Applicants believe they are the exclusive owners of the Intellectual Property listed on the Intellectual Property Schedule, they cannot be certain given the serious allegations in relation to the misappropriation of intellectual property during the CCAA proceedings that Mr. Shaw or Mrs. Shaw will not later assert an interest in such Intellectual Property that would diminish the value of the restructured Business and constitute a collateral attack on the intended purpose of the AVO to convey the Intellectual Property (and all other Purchased Assets) to the Purchaser on a "free and clear" basis. The Transaction, and the purchase price that the Purchaser is prepared to pay, is premised on the Purchaser acquiring the Intellectual Property free and clear of any competing claims by Mr. Shaw or Mrs. Shaw. Accordingly, the granting of the proposed declaration, on notice to Mr. Shaw and Mrs. Shaw, is necessary and appropriate to protect the integrity of the Intellectual Property listed on the Intellectual Property schedule, facilitate the completion of the Transaction, and maximize value for the benefit of the Applicants and their stakeholders. The Monitor supports the granting of the declaration.³³

No Exercise of Right or Remedies

37. The proposed AVO contains a waiver of default provision (the "Waiver Provision") that provides that any Person that is a party any Agreement (a) that constitutes a Purchased Asset or Purchased Business Name and is transferred to the Purchaser at Closing (an "Assigned Agreement"), or (b) to which any Purchased Subsidiary is a party as of Closing (a "Purchased Subsidiary Agreement") is forever barred from exercising any right or remedy under such

³¹ Re Taber Water Disposal Inc., 2024 ABKB 680 at paras. 31 and 34.

³² Re Golden Band Resources Inc., 2018 SKQB 284 at paras. 24-27 and 82.

³³ Third Report at para. 46.

15

Agreement by reason of the insolvency of the Applicants, the commencement of the CCAA proceedings or NOI proceedings, the completion of the Transaction, any assignment or change of control occurring in connection with the Transaction, and any default under such Agreement that is not continuing after Closing.

- 38. The purpose of the Waiver Provision is to protect the integrity of the Agreements that are being assigned to the Purchaser or retained by the Purchased Subsidiaries, preserve the stability and going-concern operations of the Business, and prevent counterparties from exercising rights based on pre-Closing events to undermine the successful completion of the Transaction. The reasons for seeking the Waiver Provision are more fully described in paragraphs 53-58 of the Third Report.
- 39. As it relates to Assigned Agreements, relief in the nature of the Waiver Provision is consistent with precedent and is commonly granted in assignment and approval and vesting orders in CCAA proceedings.³⁴
- 40. As it relates to the Purchased Subsidiary Agreements, the Waiver Provision is necessary to preserve the value of the Purchased Subsidiaries and the international Business being acquired by the Purchaser. Despite the fact that the Purchased Subsidiaries are not Applicants in the CCAA Proceeding (and as such the Purchaser cannot directly acquire their underlying assets), the sale of the Equity Interests of the Purchased Subsidiaries is an important element of the overall Transaction and will generate significant value for the benefit of the Applicants and their stakeholders. The Waiver Provision in relation to the Purchased Subsidiary Agreements is particularly important in the circumstances of this case, where the Purchaser is agreeing to purchase the shares of foreign subsidiaries without the benefit of local insolvency or recognition proceedings. The Monitor also notes that Tim Shaw (or companies controlled by him) are parties

³⁴ In the Matter of a Plan of Compromise or Arrangement of Stantive Technologies Group Inc. et al, Approval and Vesting Order of Justice Penny dated February 25, 2019 [CV-19-00614838-00CL] at para 10(b); In the Matter of a Plan of Compromise or Arrangement of Nordstrom Canada Retail, Inc. et al, Approval and Vesting Order of Chief Justice Morawetz dated February 17, 2023 [CV-23-00695619-00CL] at para 4(b); In the Matter of a Plan of Compromise or Arrangement of Essar Steel Algoma Inc. et al, Assignment Order of Justice Hainey (as he then was) dated October 17, 2018 [CV-15-000011169-00CL] at para 6; In the Matter of a Plan of Compromise or Arrangement of LoyaltyOne, Co., Assignment Order of Justice Conway dated May 12, 2023 [CV-23-00696017-00CL] at para 3(b); In the Matter of a Plan of Compromise or Arrangement of MAV Beauty Brands Inc. et al, Assignment Order of Justice Cavanagh dated November 24, 2023 [CV-23-00709610-00CL] at para 3(c); In the Matter of a Plan of Compromise or Arrangement of Gesco industries Inc. et al, Assignment Order of Justice Steele dated June 21, 2023 [CV-19-00699824-00CL] at para 8.

to a number of agreements with Purchased Subsidiaries which are important to the ongoing operations of those Purchased Subsidiaries.

- 41. There is precedent in the context of corporate restructurings for the Court to grant relief similar to the Waiver Provision to protect and preserve the value of contracts held by non-applicants.
- 42. In the *SkyLink Aviation* CCAA proceedings, Justice Morawetz (as he was then) granted a plan sanction order that prohibited contractual counterparties from exercising rights or remedies against the non-applicant "SkyLink Companies" operating in a number of foreign jurisdictions (including Kenya, United Arab Emirates and Sudan) by virtue of various specified defaults, including as a result of the insolvency of the CCAA applicant, the completion of the CCAA plan, or any event that occurred prior to the implementation of the CCAA plan that would have entitled the counterparty to exercise rights or remedies.³⁵
- 43. In the *Cannabist* CBCA proceedings, this Court recently granted a final order approving a CBCA plan of arrangement which included similar waiver provisions in respect of agreements held by certain non-applicant "CBST Entities".³⁶ The Court noted that the purpose of the waiver provision was to prevent creditors from undermining the purpose of the restructuring (in that case, a CBCA arrangement) through the exercise of rights or remedies.³⁷
- 44. The Applicants respectfully submit that the granting of the Waiver Provision in relation to the Assigned Agreements and the Purchased Subsidiary Agreements is warranted in this case to facilitate an orderly and effective transition of the Business to the Purchaser and prevent counterparties from frustrating the successful completion of the Transaction.

B. The Court Should Approve the Extended Stay Period

45. The Stay of Proceedings currently expires on August 1, 2025, which was the period intended to allow for finalizing the definitive agreement with a successful bidder and return to Court for approval of a proposed transaction. The Ancillary Order seeks to extend the Stay of Proceedings up to and including September 5, 2025, to allow for the closing of the Transaction

³⁵ In the Matter of a Plan of Compromise and Arrangement of SkyLink Aviation Inc., Plan Sanction Order dated April 23, 2013 [Court File No. 13-1003300-CL] at para. 28.

³⁶ In the Matter of a Proposed Arrangement of the Cannabist Company Holdings (Canada) Inc. and 16834434 Canada Inc., Final Order dated May 21, 2025 [CV-25-00739458-00CL] at para. 10 (attached hereto at Schedule A).

³⁷ Re Cannabist Company Holdings (Canada) Inc. et al., 2025 ONSC 3004 at paras. 82-84.

and to prepare the materials needed to seek approval of, among other things, the distribution of proceeds derived from the Transaction and the possible termination of the CCAA Proceeding.

- 46. Section 11.02(2) of the CCAA gives this Court the authority to grant an extension of the stay of proceedings for any period "it considers necessary".³⁸ To do so, this Court must be satisfied that circumstances exist that make the order appropriate and that the Applicants have acted, and are acting, in good faith and with due diligence.³⁹ A stay of proceedings is appropriate to provide a debtor with breathing room while it seeks to emerge from the CCAA.⁴⁰
- 47. The Applicants have acted and are continuing to act in good faith and with due diligence. Since the granting of the Second Stay Extension Order, the Applicants have, among other things, engaged with the Purchaser on the implementation of the Transaction, communicated with the Applicants' employees and stakeholders to keep them informed about the progress of the CCAA Proceeding, and stabilized their going-concern operations.⁴¹
- 48. The Applicants believe the Extended Stay Period is necessary and appropriate in the circumstances. The requested extension of the Stay Period will provide the Applicants with the breathing space and operational stability to continue to preserve the Business as a going concern while maximizing value for the benefit of their stakeholders through the implementation of a transaction with the Successful Bidder.
- 49. With the assistance of the Monitor, the Applicants have prepared a nine-week revised cash flow forecast for the period ending the week of September 5, 2025 (the "**Revised Cash Flow Forecast**").⁴²
- 50. The Revised Cash Flow Forecast demonstrates that the Applicants have sufficient liquidity to operate through the proposed Extended Stay Period, subject to the approval of the Third Amendment and the corresponding increase of the DIP Lender's Charge.⁴³
- 51. The Monitor is both supportive of the proposed Extended Stay Period, as is the DIP Lender if the Third Amendment is approved.⁴⁴

³⁸ CCAA, s. 11.02(2).

³⁹ CCAA, s. 11.02(3).

⁴⁰ Ted Leroy Trucking [Century Services] Ltd (Re), 2010 SCC 60 at para 14.

⁴¹ Fourth Hustrulid Affidavit at para. 43.

⁴² Fourth Hustrulid Affidavit at para. 45.

⁴³ Supplemental to the Third Report of the Monitor dated July 17, 2025 at para 12.

⁴⁴ *Ibid* at para. 17(d).

C. The Court Should Approve the Third Amendment and Increase the DIP Lender's Charge

- 52. Pursuant to Initial Order, this Court approved the Applicants' Amended DIP Term Sheet and granted a corresponding DIP Lender's Charge in the maximum principal amount of \$1,836,000 plus interest and fees. 45 As per the Stay Extension Order, this Court approved the first amendment to the DIP Facility and increased the corresponding DIP Lender's Charge to the maximum principal amount of \$2,626,500..46 As per the Second Stay Extension Order, this Court approved the Applicants' Second Amendment and increased the corresponding DIP Lender's Charge to the maximum principal amount of \$3,646,500.47 The Applicants are now seeking to increase the DIP Facility to the maximum amount of \$4,641,000 and approve the corresponding increase of the DIP Lender's Charge.
- 53. Section 11.2 of the CCAA permits the Court to grant the Second Amended DIP Facility and the DIP Lender's Charge on notice to those secured creditors that would be affected and in an amount that the Court considers appropriate having regard to the Applicants' cash flow forecast.48
- 54. All secured creditors who are affected by the proposed DIP Lender's Charge, including the increase thereof, have been served with a copy of the Applicants' Motion Record. 49
- To date, the Court has approved the Amended DIP Term Sheet and the First and Second 55. Amendments and granted the DIP Lender's Charge in the maximum amount of \$3,646,500, taking into account the factors set out at subsection 11.2(4) of the CCAA.⁵⁰ The Applicants submit that factors under subsection 11.2(4) of the CCAA continue to be satisfied under the circumstances, and support the request for an extension of the borrowings under the Second Amendment and a corresponding increase to the DIP Lender's Charge.⁵¹

⁴⁵ Fourth Hustrulid Affidavit, Exhibit "B", Endorsement of Justice J. Dietrich dated May 13, 2025, at paras 34-38.

⁴⁶ Fourth Hustrulid Affidavit, Exhibit "E", Endorsement of Justice J. Dietrich dated June 27, 2025 ["Dietrich June 27 Endorsement"] at paras. 11 and 14.

⁴⁷ Fourth Hustrulid Affidavit, Exhibit "E", Second Stay Extension Order of Justice J. Dietrich dated June 27, 2025 at <u>para 14.</u>
⁴⁸ CCAA, <u>s. 11.2(1)</u>.

⁴⁹ Affidavit of Service of Levi Rivers sworn July 15, 2025.

⁵⁰ Dietrich June 27 Endorsement at paras. 12-14.

⁵¹ CCAA, <u>s. 11.2(4)</u>.

- 56. The Revised Cash Flow Forecast shows that the Applicants require access to the Third Amended DIP Facility to provide the Applicants with necessary funding to continue their Business and operations and to advance their restructuring efforts.
- 57. The Monitor supports the Third Amendment to the Amended DIP Term Sheet by the Applicants and the corresponding increase of the DIP Lender's Charge. The Revised Cash Flow Forecast demonstrates that at least a further \$1,000,000 will need to be advanced under the Third Amendment to fund the Applicants' working capital needs over the Extended Stay Period.

D. The Court Should Approve the Sealing Provision

- 58. The Applicants seek an order sealing the Confidential Exhibit, being an unredacted copy of the Asset Purchase Agreement, and the Confidential Supplement, which contains a summary of the bids received throughout the SISP until further order of this Court.
- 59. Pursuant to subsection 137(2) of the *Courts of Justice Act*, this Court has the jurisdiction to order that any document filed in a civil proceeding be treated as confidential, sealed and not form part of the public record.⁵²
- 60. The test for a sealing order was established by the Supreme Court of Canada in *Sierra Club*⁵³ and subsequently in *Sherman Estate*.⁵⁴ The test involves three prerequisites which must be satisfied:
 - (a) whether court openness poses a serious risk to an important public interest;
 - (b) whether the order sought is necessary to prevent this serious risk to the identified interest because reasonable alternative measure will not prevent this risk; and
 - (c) whether, as a matter of proportionality, the benefits of the order outweigh its negative effects.⁵⁵

⁵² CJA, supra s. 137(2).

⁵³ Sierra Club of Canada v. Canada (Minister of Finance), 2002 SCC 41.

⁵⁴ Sherman Estate v. Donovan, <u>2021 SCC 25 (CanLII)</u>, [2021] 2 SCR 75.

⁵⁵ Ibid at para 38.

- 61. The requirements set forth in *Sherman Estate* are satisfied:
 - (a) public disclosure of the Confidential Exhibit and the Confidential Supplement poses a serious and immediate risk to the interests of stakeholders. In particular, disclosure of the Purchaser's valuation of the Business prior to the closing of the Transaction would be highly prejudicial to any future marketing efforts that may become necessary if the Transaction does not close. Such disclosure would compromise the integrity of the SISP, discourage alternative bids, and significantly impair the Applicants' ability to maximize value for the benefit of stakeholders;⁵⁶
 - (b) the sealing order sought is necessary to mitigate this risk. No reasonable alternative measures exist that would adequately protect the commercially sensitive information contained in the Confidential Exhibit and Confidential Supplement. The scope of the sealing provision is appropriately limited and remains subject to further order of this Court; and
 - (c) the benefits of granting the sealing provision outweigh any potential negative effects. The proposed relief protects the confidentiality of commercially sensitive information during a critical stage of the restructuring process, while preserving the transparency of the proceedings to the greatest extent possible under the circumstances.

E. The Court Should Approve the Monitor's Activities in the Reports and the Professional Fees of the Monitor

(1) Monitor's Reports and Activities

- 62. In *Re Target Canada Co.*, Morawetz R.S.J. (as he then was) stated that a request to approve a monitor's report "is not unusual"⁵⁷ and that: there are good policy and practical reasons for the court to approve of Monitor's activities and providing a level of protection for Monitors during the CCAA Proceeding.
- 63. In this case, the Monitor's Reports, and the conduct and activities of the Monitor described therein should be approved. The Monitor has acted reasonably and carried out its activities in a

⁵⁶ Fourth Hustrulid Affidavit at para. 50.

⁵⁷ Re Target Canada Co., 2015 ONSC 7574 at para 2.

manner consistent with the CCAA and in compliance with the Initial Order. No party has put evidence to the contrary.

(2) Monitor's Professional Fees

- 64. Pursuant to paragraph 34 of the Initial Order, this Court has jurisdiction to approve the accounts of FTI, in its capacity as Monitor, and its legal counsel, which provides that the Monitor and its legal counsel shall pass their accounts from time to time before a judge of this Court.⁵⁸
- 65. The test on a motion to pass accounts is to consider the "overriding principle of reasonableness".⁵⁹ The overall value contributed by the Monitor and its counsel is the predominate consideration in assessing the reasonableness of the accounts.⁶⁰
- 66. As the Court of Appeal for Ontario held in *Bank of Nova Scotia v Diemer*, this Court does not undertake a line-by-line analysis of the invoices. Rather, the guiding principles on fee approvals of this nature is whether the fees are fair, reasonable, and proportionate given the value of the Applicants' assets and liabilities, as well as the complexity of the Applicants' Business and the restructuring proceeding.⁶¹
- 67. The Monitor, with the assistance of Stikeman, carried out extensive activities during the times subject to the fee affidavits, as detailed in the Monitor's Reports. The more significant responsibilities that the Monitor has assumed include: (a) implementing the SISP; (c) monitoring the Business; and (d) negotiating the Asset Purchase Agreement. The Monitor also assisted and, in many cases, dealt directly with suppliers, creditors, and other stakeholders to maintain normal course operations following the commencement of the CCAA Proceeding.
- 68. The time spent, and thus the fees and disbursements of the Monitor and its legal counsel resulting from their activities, are commensurate with the significant role and responsibilities and activities undertaken. The work has been undertaken with a view to advancing the interests of the Applicant and its stakeholders.

⁵⁸ Fourth Hustrulid Affidavit, Exhibit "B", Initial Order, para. 34.

⁵⁹ Nortel Networks Inc, 2022 ONSC 6680 at para 10.

⁶⁰ Re Nortel Networks Corporation et al, 2017 ONSC 673 at paras. 15 and 21.

⁶¹ Bank of Nova Scotia v. Diemer, 2014 ONCA 851 at para. 33.

- 69. The Monitor and Stikeman's Professional Fees are comparable of similar services regarding significant and complex commercial restructuring matters.
- 70. Accordingly, it is respectfully submitted that a consideration of the factors articulated by the courts support the conclusion that the remuneration of the Monitor and its legal counsel are fair and reasonable and their fees and disbursements should be approved.

PART V – RELIEF REQUESTED

71. Based on the foregoing, the Applicants request the AVO, substantially in the form appended at tab 3 to the Applicants' motion record, and the Ancillary Order, substantially in the form appended at tab 5 of the Applicants' motion record.

PURSUANT TO RULE 4.06(2.1), THE UNDERSIGNED certifies that they are satisfied as to the authenticity of every authority cited in this factum.

Simran Joshi
Simran Joshi (LSO #89775A)

ALL OF WHICH IS RESPECTFULLY SUBMITTED THIS 17TH DAY OF JULY, 2025.

/s/ Reconstruct

RECONSTRUCT LLP

SCHEDULE "A"

List of Authorities

- 1. Re Nortel Networks Corp. (2009), 2009 CanLII 39492 (ON SC)
- 2. Re Brainhunter Inc. Re, 2009 CanLII 67659 (ON SC)
- 3. Re Canwest Publishing Inc./Publications Canwest Inc., 2010 ONSC 2870
- 4. Re Nelson Education Limited, 2015 ONSC 5557
- 5. Re Bloom Lake, 2015 QCCS 1920
- 6. Target Canada Co. (Re), 2015 ONSC 1487
- 7. Royal Bank of Canada v Soundair Corp., 1991 ONCA 2727
- 8. S.A. v. Metro Vancouver Housing Corp., 2019 SCC 4 (CanLII), [2019] 1 SCR 99
- 9. Nortel Networks v. Ernst & Young Inc., 2024 ONSC 6386 2024 CarswellOnt 18306
- 10. Re Taber Water Disposal Inc., 2024 ABKB 680
- 11. Re Golden Band Resources Inc., 2018 SKQB 284
- 12. In the Matter of a Plan of Compromise or Arrangement of Stantive Technologies Group Inc. et al, Approval and Vesting Order of Justice Penny dated February 25, 2019 [CV-19-00614838-00CL]
- 13. In the Matter of a Plan of Compromise or Arrangement of Nordstrom Canada Retail, Inc. et al, Approval and Vesting Order of Chief Justice Morawetz dated February 17, 2023 [CV-23-00695619-00CL]
- 14. In the Matter of a Plan of Compromise or Arrangement of Essar Steel Algoma Inc. et al, Assignment Order of Justice Hainey (as he then was) dated October 17, 2018 [CV-15-000011169-00CL]
- 15. In the Matter of a Plan of Compromise or Arrangement of LoyaltyOne, Co., Assignment Order of Justice Conway dated May 12, 2023 [CV-23-00696017-00CL]
- In the Matter of a Plan of Compromise or Arrangement of MAV Beauty Brands Inc. et al, Assignment Order of Justice Cavanagh dated November 24, 2023 [CV-23-00709610-00CL]
- 17. In the Matter of a Plan of Compromise or Arrangement of Gesco industries Inc. et al, Assignment Order of Justice Steele dated June 21, 2023 [CV-19-00699824-00CL]
- 18. In the Matter of a Plan of Compromise and Arrangement of SkyLink Aviation Inc., Plan Sanction Order dated April 23, 2013 [Court File No. 13-1003300-CL]
- 19. In the Matter of a Proposed Arrangement of the Cannabist Company Holdings (Canada) Inc. and 16834434 Canada Inc., Final Order dated May 21, 2025 [CV-25-00739458-00CL] at para. 10 (attached hereto at Schedule A).

20. Re Cannabist Company Holdings (Canada) Inc. et al., 2025 ONSC 3004
21. Ted Leroy Trucking [Century Services] Ltd (Re), 2010 SCC 60
22. Sierra Club of Canada v. Canada (Minister of Finance), 2002 SCC 41
23. Sherman Estate v. Donovan, 2021 SCC 25 (CanLII), [2021] 2 SCR 75
24. Re Target Canada Co., 2015 ONSC 7574
25. Nortel Networks Inc, 2022 ONSC 6680
26. Re Nortel Networks Corporation et al, 2017 ONSC 673
27. Bank of Nova Scotia v Diemer, 2014 ONCA 851

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

THE HONOURABLE)	WEDNESDAY, THE 21st
JUSTICE DIETRICH)	DAY OF MAY, 2025

IN THE MATTER OF AN APPLICATION UNDER SECTION 192 OF THE CANADA BUSINESS CORPORATIONS ACT, R.S.C. 1985, C. C-44, AS AMENDED

AND IN THE MATTER OF RULES 14.05(2) AND 14.05(3) OF THE RULES OF CIVIL PROCEDURE

AND IN THE MATTER OF A PROPOSED ARRANGEMENT OF THE CANNABIST COMPANY HOLDINGS (CANADA) INC. AND 16834434 CANADA INC., AND INVOLVING THE CANNABIST COMPANY HOLDINGS INC, PATRIOT CARE CORP., CURATIVE HEALTH LLC, COLUMBIA CARE DC LLC, MISSION BAY, LLC, CCUT PHARMACY LLC, COLUMBIA CARE PENNSYLVANIA LLC, COLUMBIA CARE INDUSTRIAL HEMP LLC, CURATIVE HEALTH CULTIVATION LLC, COLUMBIA CARE NY LLC, FOCUSED HEALTH LLC, COLUMBIA CARE NEW JERSEY LLC, COLUMBIA CARE WV INDUSTRIAL HEMP LLC, CCPA INDUSTRIAL HEMP LLC, CC OH REALTY LLC, CCF HOLDCO LLC, CC CALIFORNIA LLC, COLUMBIA CARE MD LLC, COLUMBIA CARE DE MANAGEMENT LLC, COLUMBIA CARE DELAWARE, LLC, AND COLUMBIA CARE LLC

FINAL ORDER

THIS APPLICATION made by The Cannabist Company Holdings (Canada) Inc. ("The Cannabist Canada Company") and 16834434 Canada Inc. ("168Co" and together with The Cannabist Canada Company, the "Applicants") for a final order (this "Final Order") in connection with an arrangement (the "Arrangement") pursuant to section 192 of the Canada Business Corporations Act, R.S.C. 1985, c. C-44, as amended, (the "CBCA") was heard this day in person.

ON READING the Notice of Application issued on March 20, 2025, the affidavits of David Hart, sworn March 21, 2025, and April 21, 2025, and the exhibits attached thereto, the affidavits of Brittney Ketwaroo sworn May 1, 2025, and May 7, 2025, and the exhibits attached thereto,

the affidavits of Paul Zogala affirmed March 26, 2025, April 9, 2025, and April 29, 2025, and the exhibits attached thereto, the factum of the Applicants dated May 7, 2025, the factum of the Ad Hoc Group of Senior Noteholders dated May 7, 2025, the factum of Murchinson Ltd. dated May 9, 2025, the reply factum of the Applicants dated May 11, 2025, the Transcript Brief dated May 6, 2025, the answers to the undertakings and under advisements on cross examination of David Hart on May 2, 2025, the answers to the undertakings and under advisements on cross examination of Paul Zogala on May 2, 2025, the Notice of Motion of Murchinson Ltd. dated May 8, 2025, the Plan of Arrangement attached as Schedule "A" to this Final Order (the "Plan") and the Circular.

ON HEARING the submissions of counsel for the Applicants, Murchinson Ltd. and counsel for certain supporting holders (the "Supporting Noteholders") of: (a) the six percent (6.0%) Senior Secured Convertible Notes due June 29, 2025 (the "2025 Notes"); (b) the nine and one half percent (9.5%) Senior Secured First Lien Notes due February 3, 2026 (the "2026 Notes"); and (c) the nine percent (9.0%) Senior Secured Convertible Notes due March 19, 2027 (the "2027 Notes", and together with the 2025 Notes and the 2026 Notes, the "Senior Notes" and holders of such Notes, the "Senior Noteholders") co-issued by The Cannabist Canada Company and The Cannabist Company Holdings Inc. ("The Cannabist Company"), each of whom are party to the Support Agreement dated February 27, 2025 (the "Support Agreement"), and on being advised that (a) the Director appointed under the CBCA (the "Director") does not consider it necessary to appear; and (b) this Final Order and the declaration of fairness included herein will be relied upon by the CBST Entities as the basis for an exemption pursuant to Section 3(a)(10) of the *United States Securities Act of 1933*, as amended, (the "U.S. Securities Act, regarding the distribution of the New Notes and New CBST Common Shares.

Definitions

1. **THIS COURT ORDERS** that all definitions used in this Final Order shall have the meanings ascribed thereto in the Plan or otherwise as specifically defined herein.

Service

- 2. **THIS COURT ORDERS** that there has been good and sufficient service, delivery and notice of the Application, the Interim Order, the Noteholders Meeting, the Circular, the Letter of Transmittal and Election Form, and the Shareholder Letter (as defined in the Interim Order) to all Persons upon which service, delivery and notice were required by the terms of the Interim Order and that the Noteholders Meeting was duly called and conducted in conformity with the Interim Order and the CBCA.
- 3. **THIS COURT ORDERS** that service of this Final Order shall be made on all Persons who appeared on this Application, either by counsel or in person, and upon the Director, but is otherwise dispensed with.

Oppression Action

4. **THIS COURT ORDERS** that the application commenced by Murchison Ltd. pursuant to sections 192 and 241(1) of the CBCA bearing court file no. CV-25-00739982-00CL is hereby dismissed.

Approval of Arrangement

THIS COURT ORDERS that:

(a) the Arrangement, as described in the Plan, is an arrangement within the meaning of section 192 of the CBCA;

- (b) the Court is satisfied that the Applicants have acted, and are acting, in good faith and with due diligence, and have complied with the provisions of the CBCA and the Interim Order in all respects;
- (c) the Arrangement, as described in the Plan, and including the distribution of New Notes, New CBST Common Shares, Early Consent Consideration, Additional Early Consent Consideration and Anti-Dilutive Warrants contemplated thereby, is fair and reasonable, and fair to all persons entitled to receive such securities; and
- (d) the Arrangement, as described in the Plan, shall be and is hereby approved.
- 6. THIS COURT ORDERS that each of the CBST Entities, the Indenture Trustee, the Warrant Agent, the Transfer Agent, the Proxy, Information and Exchange Agent, CDS, Intermediaries and other applicable agents are authorized and directed to take all steps and actions necessary or appropriate to implement the Plan and the Arrangement and the other transactions contemplated thereby in accordance with and subject to the terms of the Plan, including (a) to enter into any agreements or other documents which are to come into effect in connection with the Arrangement, and (b) to execute and deliver (or direct to be executed and delivered) such releases, terminations, discharges, security agreements, liens and guarantees as are required to give effect to the Plan and the Arrangement.
- 7. **THIS COURT ORDERS** that as of the Effective Date, and as at the times and sequences set forth in the Plan, the Plan and all associated steps and transactions shall be binding and effective as set out in the Plan, and on the terms and conditions set forth in this Final Order, upon (a) the Applicants, CBST, the Existing Guarantors, the New Guarantors and the other CBST Entities; (b) all current and former Senior Noteholders and CBST Shareholders; (c) all current and former beneficial holders of Senior Notes, CBST Common Shares and CBST

Proportionate Voting Shares; (d) the Indenture Trustee; (e) the Warrant Agent; (f) the Transfer Agent; (g) the Depositary; (h) CDS; (i) all holders of Released Claims; (j) the Released Parties; and (k) all other Persons affected by the Plan.

- 8. **THIS COURT ORDERS** that the transactions contemplated by and to be implemented pursuant to the Plan shall not be void or voidable under federal, provincial, or similar legislation and shall not constitute and shall not be deemed to be preferences, assignments, fraudulent conveyances, transfers at undervalue, or other reviewable transactions under any applicable federal, provincial or similar legislation relating to preferences, assignments, fraudulent conveyances or transfers at undervalue.
- 9. **THIS COURT ORDERS** that from and after the Effective Date any conflict between (i) the Plan, and (ii) the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, agreement, credit agreement, financing agreement, mortgage, security agreement, indenture, trust indenture, note, loan agreement, guarantee, inter-creditor agreement, subordination agreement, commitment letter, agreement for sale, lease, license or other legally binding arrangement, written or oral and any and all amendments or supplements thereto (each an, "**Agreement**") existing between any Person and any of the CBST Entities as at the Effective Time, will be deemed to be governed by the terms, conditions and provisions of the Plan and this Final Order, which shall take precedence and priority.

No Default

10. **THIS COURT ORDERS** that from and after the Effective Date, all Persons shall be deemed to have:

- (a) waived any and all defaults or events of default, accelerations, third-party change of control rights or any non-compliance with any covenant, warranty, representation, term, provision, condition or obligation, expressed or implied, in any Agreement, arising out of, or in connection with, the Senior Notes, the Note Claims, the Note Documents, the Plan, the Arrangement, the Support Agreement, the transactions contemplated hereunder or thereby, the CBCA Proceedings and any other proceedings commenced with respect to or in connection with the Plan and any and all amendments or supplements thereto. Any and all notices of default and demands for payment or any action, right, step or proceeding taken, exercised or commenced in connection with any of the foregoing shall be deemed to have been rescinded, irrevocably waived and of no further force or effect; provided that nothing shall be deemed to excuse the Applicants or the other CBST Entities, and their respective successors and assigns, from performing their obligations under the Plan, the New Notes Documents or any other contract or agreement entered into pursuant to, in connection with, or contemplated by, the Plan; and
- (b) agreed that if there is any conflict between the provisions of any Agreement existing between such Person and the Applicants, CBST or the other CBST Entities prior to the Effective Date and the provisions of the Plan, then the provisions of the Plan take precedence and priority and the provisions of such Agreement are deemed to be amended accordingly.

Releases and Injunctions

11. **THIS COURT ORDERS** that, as of the Effective Date, at the time and in the sequence, as applicable, set forth in the Plan, the releases and injunctions set forth in Article 4 of the Plan shall be and are hereby approved and are binding and effective as set out in the Plan.

Extra-Territorial Assistance

12. **THIS COURT HEREBY REQUESTS** the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada, the United States, or any other foreign jurisdiction, to give effect to this Final Order and to act in aid of and to assist this Court and the Applicants and their respective agents in carrying out the terms of this Final Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Applicants and their respective agents as may be necessary or desirable to give effect to this Final Order, to grant representative status to the Applicants in any foreign proceeding, or to assist the Applicants and its respective agents in carrying out the terms of this Final Order.

Indenture Trustee

13. **THIS COURT ORDERS** that the Indenture Trustee, the Warrant Agent, the Transfer Agent, the Depositary and CDS shall not incur any liability as a result of carrying out or observing the provisions of this Final Order, the Plan, the Arrangement or the taking of any action incidental hereto, save and except for any gross negligence or wilful misconduct on its part.



Schedule "A"

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

IN THE MATTER OF AN APPLICATION UNDER SECTION 192 OF THE CANADA BUSINESS CORPORATIONS ACT, R.S.C. 1985, c. C-44, AS AMENDED, AND RULES 14.05(2) AND 14.05(3) OF THE RULES OF CIVIL PROCEDURE

AND IN THE MATTER OF A PROPOSED ARRANGEMENT OF THE CANNABIST COMPANY HOLDINGS (CANADA) INC. AND 16834434 CANADA INC. INVOLVING THE CANNABIST COMPANY HOLDINGS INC., PATRIOT CARE CORP., CURATIVE HEALTH LLC, COLUMBIA CARE DC LLC, MISSION BAY, LLC, CCUT PHARMACY LLC, COLUMBIA CARE PENNSYLVANIA LLC, COLUMBIA CARE INDUSTRIAL HEMP LLC, CURATIVE HEALTH CULTIVATION LLC, COLUMBIA CARE NY LLC, FOCUSED HEALTH LLC, COLUMBIA CARE NEW JERSEY LLC, COLUMBIA CARE WV INDUSTRIAL HEMP LLC, CCPA INDUSTRIAL HEMP LLC, CC OH REALTY LLC, CCF HOLDCO LLC, CC CALIFORNIA LLC, COLUMBIA CARE MD LLC, COLUMBIA CARE DE MANAGEMENT LLC, COLUMBIA CARE DELAWARE, LLC, AND COLUMBIA CARE LLC

PLAN OF ARRANGEMENT

May 21, 2025

TABLE OF CONTENTS

		<u>Page</u>
ARTICLE 1 IN	ITERPRETATION	1
1.1	Definitions	1
1.2	Certain Rules of Interpretation	
1.3	Governing Law	
1.4	Currency	
1.5	Date for Any Action	
1.6	Time	
1.7	Binding Effect	
	HE ARRANGEMENT	
2.1	Corporate Authorizations	_
2.1	Effective Date Transactions	
2.2	2027 Noteholder Election Mechanics	
2.4	Other Implementation Steps	
2.5	Fractional Interests	
2.6	Rounding of Cash	
2.7	Calculations	15
	XCHANGE OF NOTES AND ISSUANCE OF NEW NOTES, NEW CBST ARES, AND ANTI-DILUTIVE WARRANTS	15
3.1	Issuance of New Notes	_
3.2	Surrender of Notes	
3.3	Issuance of New CBST Common Shares	
3.4	Issuance of Anti-Dilutive Warrants	
3.5	Payment of the Early Consent Consideration	
3.6	No Liability in Respect of Deliveries	16
3.7	Deposit of Notes	
3.8	Letter of Transmittal and Election Form	18
3.9	Lost Certificates	
3.10	Securities Law Matters	18
3.11	Withholding Rights	18
ARTICLE 4 R	ELEASES	19
4.4	Release of Released Parties	40
4.1		
4.2	Injunctions	20
ARTICLE 5 C	ONDITIONS PRECEDENT AND IMPLEMENTATION	20
5.1	Conditions to Plan Implementation	20
5.2	Waiver of Conditions	
5.3	Effectiveness	
5.4	Effect of Non-Occurrence of Effective Date	
_	ENERAL	
ANTICLE 0 G		
6.1	Deemed Consents, Waivers and Agreements	21
6.2	Waiver of Defaults	
6.3	Compliance with Deadlines	
6.4	Paramountcy	22
6.5	Deeming Provisions	23

6.6	Amendment or Modification of Plan	23
6.7	Notices	23
6.8	Consent of Supporting Noteholders	25
	Further Assurances	

ARTICLE 1 INTERPRETATION

1.1 Definitions

In this Plan, unless otherwise stated:

"2025 Notes" means the 6.0% Senior Secured Convertible Notes due June 29, 2025, co-issued by CBST and CBST Canada 1 pursuant to the Existing Indenture;

"2025 Noteholder" means a Holder of the 2025 Notes;

"2026 Notes" means the 9.5% Senior Secured First-Lien Notes due February 3, 2026, co-issued by CBST and CBST Canada 1 pursuant to the Existing Indenture;

"2026 Noteholder" means a Holder of the 2026 Notes;

"2027 Notes" means the 9.0% Senior Secured Convertible Notes due March 19, 2027, co-issued by CBST and CBST Canada 1 pursuant to the Existing Indenture;

"2027 Noteholder" means a Holder of the 2027 Notes;

"Additional Amounts" has the meaning given thereto in Section 3.11(b);

"Additional Early Consent Consideration" means consideration in the aggregate amount of \$1,500,000, payable by CBST in cash at maturity of the New Senior Notes, except that such payment will be accelerated and made upon the occurrence of certain milestones, as follows: (i) \$750,000 shall be paid on the date on which CBST has received aggregate proceeds for Approved Sales equal to or greater than \$15 million; and (2) \$750,000 shall be paid on the date on which CBST has received aggregate proceeds for Approved Sales equal to or greater than \$20 million, with, for greater certainty, any balance remaining payable at maturity of the New Senior Notes;

"Amalgamation" has the meaning given thereto in Section 2.2(h);

"Amended and Restated Trust Indenture" means the Amended and Restated Trust Indenture, substantially in the form appended to the Circular, to be entered into among CBST, CBST Canada 1, the Existing Guarantors, the New Guarantors and the Indenture Trustee on the Effective Date, which shall amend and restate the Existing Indenture and govern the New Notes upon this Plan becoming effective on the Effective Date and includes for the purpose of this Plan, the First Supplemental Indenture;

"Anti-Dilutive Warrants" means 118,246,947 warrants to purchase CBST Common Shares for a period of two years from the Effective Date at an exercise price per share of C\$0.14, issued on the Effective Date pursuant to the Warrant Agency Agreement and this Plan to existing CBST Shareholders as of the Warrant Record Date;

"Applicants" means CBST Canada 1 and CBST Canada 2;

"**Approved Sales**" means those certain pre-approved asset sales disclosed by CBST and CBST Canada 1 to the Supporting Noteholders in writing;

- "Arrangement" means the arrangement under section 192 of the CBCA on the terms and subject to the conditions set out in this Plan, subject to any amendments, modifications and/or supplements made thereto in accordance with the terms of this Plan or made at the discretion of the Court in the Final Order with the prior written consent of the Applicants, CBST, and the Requisite Supporting Noteholders, each acting reasonably;
- "Articles of Arrangement" means the articles of arrangement of the Applicants in respect of the Arrangement, in form and substance satisfactory to the Applicants and the Requisite Supporting Noteholders, that are required to be filed with the CBCA Director in order for the Arrangement to become effective on the Effective Date;
- "Business Day" means any day, other than a Saturday, Sunday or a statutory or civic holiday, on which banks are generally open for business in Toronto, Ontario and New York, New York;
- "Canadian Securities Administrators" means, collectively, the applicable securities commissions or regulatory authorities in each of the provinces and territories of Canada:
- "Canadian Securities Laws" means, collectively, and, as the context may require, the applicable securities Laws of each of the provinces and territories of Canada, and the respective regulations and rules made under those securities Laws together with all applicable instruments, blanket orders and rulings of the Canadian Securities Administrators and all discretionary orders or rulings, if any, of the Canadian Securities Administrators made in connection with the transactions contemplated by the Plan, as the context may require;
- "CBCA" means the Canada Business Corporations Act, R.S.C. 1985, c. C-44, as amended;
 - "CBCA Director" means the Director appointed under section 260 of the CBCA;
- "CBCA Proceedings" means the proceedings commenced by the Applicants under the CBCA on March 20, 2025 in connection with this Plan;
- "CBST" means The Cannabist Company Holdings Inc., a company existing under the laws of British Columbia;
- "CBST Amalco" means The Cannabist Company Holdings (Canada) Inc., the corporation that will result from the Amalgamation under the federal laws of Canada;
- "CBST Canada 1" means The Cannabist Company Holdings (Canada) Inc., a corporation existing under the federal laws of Canada, and a direct, wholly-owned subsidiary of CBST;
- "CBST Canada 2" means 16834434 Canada Inc., a corporation existing under the federal laws of Canada, and a direct, wholly-owned subsidiary of CBST;
 - "CBST Common Shares" means common shares in the capital of CBST;

"CBST Entities" means, collectively, the Applicants, CBST, and each other direct or indirect subsidiary of CBST, and "CBST Entity" means any of them;

"CBST Proportionate Voting Shares" means proportionate voting shares in the capital of CBST;

"CBST Shareholders" means the Holders of CBST Shares;

"CBST Shares" means, collectively, the CBST Common Shares and CBST Proportionate Voting Shares;

"CDS" means CDS Clearing and Depository Services Inc. and its nominees, successors and assigns;

"Certificate of Arrangement" means the certificate giving effect to the Arrangement, to be issued by the CBCA Director pursuant to section 192(7) of the CBCA upon receipt of the Articles of Arrangement in respect of the Applicant in accordance with section 262 of the CBCA:

"Circular" means the management information circular of CBST and CBST Canada 1 dated March 28, 2025, including all appendices thereto, as it may be amended, modified and/or supplemented from time to time, subject to the terms of the Interim Order or other Order of the Court;

"Claim" means any right or claim of any Person that may be asserted or made in whole or in part against the applicable Persons, or any of them, in any capacity, whether or not asserted or made, in connection with any indebtedness, liability or obligation of any kind whatsoever, and any interest accrued thereon or costs payable in respect thereof, whether at Law or in equity, including by reason of the commission of a tort (intentional or unintentional), by reason of any breach of contract or other agreement (oral or written), by reason of any breach of duty (including, any legal, statutory, equitable or fiduciary duty). by reason of any right of setoff, counterclaim or recoupment, or by reason of any equity interest, right of ownership of or title to property or assets or right to a trust or deemed trust (statutory, express, implied, resulting, constructive or otherwise), and together with any security enforcement costs or legal costs associated with any such claim, and whether or not any indebtedness, liability or obligation is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, undisputed, legal, equitable, secured, unsecured, perfected, unperfected, present or future, known or unknown, by guarantee, warranty, surety or otherwise, and whether or not any right or claim is executory or anticipatory in nature, including any claim made or asserted against the applicable Persons, or any of them, through any successor, assignee, affiliate, subsidiary, associated or related Person, or any right or ability of any Person to advance a claim for an accounting, reconciliation, contribution, indemnity, restitution or otherwise with respect to any matter, grievance, action (including any class action or proceeding before an administrative or regulatory tribunal), cause or chose in action, whether existing at present or commenced in the future;

"Court" means the Ontario Superior Court of Justice (Commercial List);

"**Depositary**" means Odyssey Trust Company or any other trust company, bank or equivalent financial institution agreed to in writing by the Applicants and the Requisite

Supporting Noteholders and appointed to carry out any of the duties of the Depositary hereunder;

"DRS Advice" means an advice issued by the Transfer Agent, the Warrant Agent or the Indenture Trustee evidencing the securities held by a Holder of CBST Common Shares or other CBST Entity securities (including, for greater certainty, the New CBST Common Shares), Anti-Dilutive Warrants, Notes and/or New Notes, as applicable, in lieu of a physical certificate;

"Early Consent Consideration" means consideration in the aggregate amount of \$1,500,000, payable by CBST either (i) in cash; or (ii) by the transfer of Verano Shares equal to such value as measured by the 30-day VWAP of the Verano Shares on the Effective Date, at the option of CBST;

"Early Consent Consideration Deadline" means 5:00 p.m. (Toronto Time) on March 10, 2025 or such later date as agreed between CBST, CBST Canada 1 and the Requisite Supporting Noteholders;

"Early Supporting Noteholders" means Supporting Noteholders as of the Early Consent Consideration Deadline, excluding New Convertible Note-Electing Noteholders;

"**Effective Date**" means the date shown on the Certificate of Arrangement issued by the CBCA Director;

"Effective Time" means such time on the Effective Date as may be specified by the Applicants as the time at which the Arrangement implementation steps set forth in Section 2.2 shall be deemed to commence;

"Election Deadline" means 5:00 p.m. (Toronto time) on the date that is five (5) Business Days prior to the Noteholders Meeting;

"Existing Guarantees" means the guarantees delivered pursuant to the Existing Indenture guaranteeing the indebtedness and obligations under the Existing Indenture and the Notes;

"Existing Guarantors" means collectively, Patriot Care Corp., Curative Health LLC, Columbia Care DC LLC, Mission Bay, LLC, CCUT Pharmacy LLC, Columbia Care Pennsylvania LLC, Columbia Care Industrial Hemp LLC, Curative Health Cultivation LLC, Columbia Care NY LLC, Focused Health LLC, Columbia Care New Jersey LLC, Columbia Care WV Industrial Hemp LLC, CCPA Industrial Hemp LLC, CC OH Realty LLC, CCF Holdco LLC, CC California LLC, Columbia Care MD LLC, Columbia Care De Management LLC, Columbia Care Delaware, LLC, and Columbia Care LLC;

"Existing Indenture" means the trust indenture dated as of May 14, 2020, among CBST, CBST Canada 1 and the Indenture Trustee, as amended;

"Final Order" means the Order of the Court approving the Arrangement under section 192 of the CBCA, which shall include such terms as may be necessary or appropriate to give effect to the Arrangement and this Plan, in form and substance satisfactory to the Applicants, CBST, and the Requisite Supporting Noteholders, as such

Order may be amended from time to time in a manner acceptable to the Applicants, CBST and the Requisite Supporting Noteholders;

"First Supplemental Indenture" means first supplemental indenture to the Amended and Restated Trust Indenture dated as of the Effective Date, which shall govern the New Convertible Notes upon this Plan becoming effective on the Effective Date;

"Governmental Entity" means any government, regulatory authority, governmental department, agency, commission, bureau, official, minister, Crown corporation, court, board, tribunal or dispute settlement panel or other Law-making organization (including any stock exchange and the Canadian Investment Regulatory Organization) or entity: (i) having or purporting to have jurisdiction on behalf of any nation, province, territory or state or any other geographic or political subdivision of any of them; or (ii) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;

"**Holder**" means a Person in whose name a note, warrant or share is registered, as the context requires;

"Indemnified Taxes" has the meaning given thereto in Section 3.11(b);

"Indenture Trustee" means Odyssey Trust Company as trustee under the Existing Indenture and the Amended and Restated Trust Indenture, and its successors and assigns;

"Interim Order" means the interim Order of the Court granted on March 28, 2025, pursuant to section 192 of the CBCA, which, among other things, approves the calling of, and the date for, the Noteholders Meeting, and as may be amended from time to time in a manner acceptable to the Applicants, CBST and the Requisite Supporting Noteholders;

"Interest Payment Date" has the meaning ascribed thereto in the Existing Indenture;

"Intermediary" means a broker, custodian, investment dealer, nominee, bank, trust company or other intermediary;

"Law" means any law, statute, constitution, treaty, convention, code, injunction, order, decree, consent decree, judgment, rule, regulation, ordinance or other pronouncement having the effect of law whether in Canada, the United States, or any other country, or any domestic or foreign state, county, province, territory, city or other political subdivision or of any Governmental Entity;

"Letter of Transmittal and Election Form" means the letter of transmittal and election form sent to registered Noteholders for use in connection with the Arrangement;

"Meeting Record Date" means March 7, 2025, as may be amended by the Applicants in accordance with the Interim Order;

"New CBST Common Shares" means a total of 118,209,105 CBST Common Shares to be issued to the 2025 Noteholders, 2026 Noteholders and the New Senior Note-Electing Noteholders;

"New Convertible Note-Electing Noteholder" means a 2027 Noteholder that has validly made, or has been deemed to have made, a New Convertible Note Election in accordance with Sections 2.3(a)(ii) or 2.3(c), respectively;

"New Convertible Notes" means new 9.0% Senior Secured Convertible Notes due December 31, 2028, to be issued by CBST and CBST Canada 1 and guaranteed by the Existing Guarantors and the New Guarantors, pursuant to the Amended and Restated Trust Indenture and this Plan, which notes will be denominated in U.S. Dollars and constitute a new series under the Amended and Restated Trust Indenture, and shall be issued to New Convertible Note-Electing Noteholders;

"New Convertible Note Election" has the meaning given thereto in Section 2.3(a)(ii);

"New Guarantees" means the new guarantees of the New Notes to be provided by the New Guarantors, including, for greater certainty, all general, collateral and other security agreements entered into by the New Guarantors;

"**New Guarantors**" means the CBST Entities other than the Applicants, CBST, the Existing Guarantors and the Unrestricted Subsidiaries;

"New Notes" means the New Senior Notes and New Convertible Notes:

"New Notes Documents" means, collectively, the Amended and Restated Trust Indenture, the New Notes, the New Guarantees and the guarantees to be provided by the Existing Guarantors pursuant to the Amended and Restated Trust Indenture, and the new security agreements (as applicable) to be entered into by the Existing Guarantors and the New Guarantors, and all other documents and agreements related thereto, in each case, in a form acceptable to the Applicants, CBST and the Requisite Supporting Noteholders;

"New Senior Note-Electing Noteholder" means a 2027 Noteholder that has validly made a New Senior Note Election in accordance with Section 2.3(a)(i);

"New Senior Notes" means new 9.25% senior notes due December 31, 2028, to be issued by CBST and CBST Canada 1 and guaranteed by the Existing Guarantors and the New Guarantors, pursuant to the Amended and Restated Trust Indenture and this Plan, which notes will be denominated in U.S. Dollars and constitute a new series under the Amended and Restated Trust Indenture, and shall be issued to 2025 Noteholders, 2026 Noteholders and New Senior Note-Electing Noteholders;

"New Senior Note Election" has the meaning given thereto in Section 2.3(a)(i):

"Notes" means the 2025 Notes, the 2026 Notes, and the 2027 Notes;

"Notes Claims" means all outstanding liabilities, debts and obligations, including without limitation principal and interest, any make whole, any prepayment, redemption or similar premiums, reimbursement obligations, fees, penalties, damages, guarantees, indemnities, costs, expenses or otherwise, and any other liabilities, debts or obligations, whether direct or indirect, absolute or contingent, known or unknown, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection

with, the Notes Documents, owing by any Person (whether as issuer, guarantor or otherwise) as at the Effective Date;

"Notes Documents" means, collectively, the Existing Indenture, the Notes, the Existing Guarantees and all security and other documentation related to the Notes;

"Noteholders" means Holders of the Notes;

"Noteholders Arrangement Resolution" means the resolution of the Noteholders, approving the Arrangement to be considered and voted upon at the Noteholders Meeting, substantially in the form attached as Appendix "A" to the Circular;

"Noteholders Meeting" means the meeting of Noteholders as of the Meeting Record Date called and held pursuant to the Interim Order for the purpose of considering and voting on the Noteholders Arrangement Resolution and to consider and vote on such other matters as may properly come before such meeting, and includes any adjournment(s) or postponement(s) of such meeting;

"Order" means any order entered by the Court in the CBCA Proceedings;

"Person" means any individual, firm, corporation, partnership, limited or limited liability partnership, limited or unlimited liability company, joint venture, fund, association, organization, trust, trustee, executor, administrator, legal personal representative, estate, group, unincorporated association or organization, Governmental Entity or any agency, instrumentality or political subdivision of a Governmental Entity, or any other entity or body, whether or not having legal status;

"Plan" means this plan of arrangement and any amendments, restatements, modifications and/or supplements hereto made in accordance with the terms hereof;

"Proxy, Information and Exchange Agent" means Carson Proxy Advisors or any other Person appointed by CBST or the Applicants for such purpose;

"Pro Rata Share" means a Person's respective proportionate share of the aggregate relevant Notes or aggregate relevant CBST Shares, as applicable, and identified and set forth in the Plan.

"Release Carve-Outs" has the meaning given thereto in Section 4.1;

"Released Claims" has the meaning given thereto in Section 4.1;

"Released Parties" means, collectively, (i) the CBST Entities and each of their respective current and former directors, officers, principals, members, affiliates, limited partners, general partners, managers of accounts or funds, fund advisors, employees, shareholders, financial and other advisors, legal counsel and agents, including the Proxy, Information and Exchange Agent and the Indenture Trustee each in their capacity as such, and (ii) the Supporting Noteholders and their respective current and former directors, officers, principals, members, affiliates, limited partners, general partners, managers of accounts or funds, fund advisors, employees, shareholders, financial advisors, legal counsel and agents, each in their capacity as such;

- "Requisite Supporting Noteholders" has the meaning ascribed thereto in the Support Agreement;
- "Support Agreement" means the support agreement entered into by, among others, CBST, CBST Canada 1 and the Supporting Noteholders in connection with this Plan on February 27, 2025;
- "Supporting Noteholders" has the meaning ascribed thereto in the Support Agreement;
- "Supporting Noteholders Advisors" means Goodmans LLP, Feuerstein Kulick LLP and Ducera Partners LLC;
- "Standstill and Lock-up Agreement" means the standstill and lock-up agreement to be entered into or deemed entered into by CBST, CBST Canada 1 and the Noteholders as of the Effective Date, substantially in the form attached as Appendix "I" to the Circular;
- "**Tax Act**" means the *Income Tax Act* (Canada) as amended and all regulations thereunder;
- "Taxes" means all income taxes, capital taxes, stamp taxes, charges to tax, withholdings, sales and use taxes, value added taxes, goods and services taxes, and all penalties, interest and other payments thereon or in respect thereof, including a payment under the Tax Act, the U.S. Code, or any other federal, provincial, territorial, state, municipal, local or foreign tax Law;
- "**Transfer Agent**" means Odyssey Trust Company, in its capacity as the transfer agent and registrar of CBST;
- "**Unrestricted Subsidiaries**" has the meaning ascribed thereto in the Amended and Restated Trust Indenture:
- "U.S. Code" means the United States Internal Revenue Code of 1986, as amended;
- "U.S. Securities Act" means the United States Securities Act of 1933, as amended;
- "**Verano Shares**" means the common shares in the capital of Verano Holdings Corp. owned by CBST;
- "**VWAP**" means the volume-weighted average price on the securities exchange maintained by Cboe Canada Inc.;
- "Warrant Agency Agreement" means the warrant agency agreement between CBST and the Warrant Agent to be entered into on the Effective Date, in form and substance satisfactory to CBST and the Requisite Supporting Noteholders, each acting reasonably;
- "Warrant Agent" means Odyssey Trust Company, in its capacity as warrant agent under the Warrant Agency Agreement; and

"Warrant Record Date" means the date that is two (2) Business Days prior to the Effective Date.

1.2 Certain Rules of Interpretation

For the purposes of this Plan:

- (a) Unless otherwise expressly provided herein, any reference in this Plan to an instrument, agreement or an order or an existing document or exhibit filed or to be filed means such instrument, agreement, order, document or exhibit as it may have been or may be amended, modified, restated or supplemented in accordance with its terms;
- (b) The division of this Plan into articles, sections, subsections, clauses and paragraphs is for convenience of reference only, and the descriptive headings of articles and sections are not intended as complete or accurate descriptions of the content thereof, none of which shall affect the construction or interpretation of this Plan:
- (c) The use of words in the singular or plural, or with a particular gender, including a definition, shall not limit the scope or exclude the application of any provision of this Plan to such Person (or Persons) or circumstances as the context otherwise permits;
- (d) The words "includes" and "including" and similar terms of inclusion shall not, unless expressly modified by the words "only" or "solely", be construed as terms of limitation, but rather shall mean "includes but is not limited to" and "including but not limited to", so that references to included matters shall be regarded as illustrative without being either characterizing or exhaustive;
- (e) Unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends;
- (f) Unless otherwise provided, any reference to a statute or other enactment of parliament, a legislature or other Governmental Entity includes all rules, regulations and blanket orders made thereunder, all amendments to or re-enactments of such statute or other enactment in force from time to time, and, if applicable, any statute or enactment that supplements or supersedes such statute or enactment;
- (g) References to a specific recital, article, section, subsection or clause shall, unless something in the subject matter or context is inconsistent therewith, be construed as references to that specific recital, article, section, subsection or clause of this Plan, whereas the terms "this Plan", "hereof", "herein", "hereto", "hereunder" and similar expressions shall be deemed to refer generally to this Plan and not to any particular recital, article, section, subsection, clause or other portion of this Plan and shall include any amended or restated Plan and any documents supplemental hereto:

- (h) Capitalized terms not otherwise defined herein shall have the meaning ascribed to them in the Existing Indenture; and
- (i) The word "or" is not exclusive.

1.3 Governing Law

This Plan shall be governed by and construed in accordance with the Laws of Ontario and the federal Laws of Canada applicable therein. All questions as to the interpretation or application of this Plan and all proceedings taken in connection with this Plan and its provisions shall be subject to the exclusive jurisdiction of the Court.

1.4 Currency

Unless otherwise stated, all references in this Plan to sums of money are expressed in, and all payments provided for herein shall be made in, U.S. Dollars.

1.5 Date for Any Action

If the date on which any action is required to be taken hereunder by a Person is not a Business Day, such action shall be required to be taken on the next succeeding day which is a Business Day.

1.6 Time

Time shall be of the essence in this Plan. Unless otherwise specified, all references to time expressed in this Plan and in any document issued in connection with this Plan mean local time in Toronto, Ontario, and any reference to an event occurring on a Business Day shall mean prior to 5:00 p.m. on such Business Day.

1.7 Binding Effect

The Arrangement will become effective at, and be binding at and after, the Effective Time on (i) the Applicants, CBST, the Existing Guarantors, the New Guarantors and the other CBST Entities, (ii) all current and former Noteholders and CBST Shareholders; (iii) all current and former beneficial holders of Notes, CBST Common Shares and CBST Proportionate Voting Shares, (iv) the Indenture Trustee, (v) the Warrant Agent, (vi) the Transfer Agent, (vii) the Depositary, and (viii) CDS.

ARTICLE 2 THE ARRANGEMENT

2.1 Corporate Authorizations

The adoption, execution, delivery, implementation and consummation of all matters contemplated under this Plan involving corporate action of any of the Applicants, CBST or the other CBST Entities will occur and be effective as of the Effective Date (or such other date as the Applicants, CBST, the other CBST Entities and the Requisite Supporting Noteholders may agree, acting reasonably), and will be authorized and approved under this Plan and by the Court, where appropriate, as part of the Final Order, in all respects and for all purposes and without any requirement, except as expressly provided herein, of further action by the Noteholders, the CBST

Shareholders, or by the directors or officers of the Applicants, CBST, the Existing Guarantors, the New Guarantors or any other CBST Entity.

2.2 Effective Date Transactions

At the Effective Time, the following events or transactions will occur, or be deemed to have occurred and be taken and effected, sequentially as set out below without any further authorization, act or formality, in each case, unless stated otherwise, effective as at five-minute intervals commencing at the Effective Time:

- (a) CBST and the Warrant Agent shall enter into the Warrant Agency Agreement and CBST shall cause to be issued, as a special distribution, to each CBST Shareholder its Pro Rata Share as of the Warrant Record Date (based on the number of CBST Shares held by such CBST Shareholder relative to the aggregate number of CBST Shares issued and outstanding, in each case as of the Warrant Record Date and assuming conversion of each CBST Proportionate Voting Share into 100 CBST Common Shares) of the Anti-Dilutive Warrants;
- (b) CBST, CBST Canada 1, the Existing Guarantors and the New Guarantors and the Indenture Trustee shall enter into the Amended and Restated Trust Indenture and the other New Notes Documents in respect of the issuance of the New Notes, which were not entered into prior to the Effective Time;
- (c) CBST shall pay all accrued but unpaid interest on the Notes since the last Interest Payment Date to, but excluding the Effective Date, in cash to the Indenture Trustee, for and on behalf of the Noteholders, and the Indenture Trustee shall pay (or cause to be paid) such interest payment to the Noteholders pursuant to standing instructions and customary practices, without abatement or rights of setoff or counterclaim of any nature;
- (d) each 2025 Noteholder, 2026 Noteholder and New Senior Note-Electing Noteholder shall: (i) be deemed to have surrendered their Notes in accordance with Section 3.2; and (ii) receive as consideration:
 - A. its respective Pro Rata Share (based on the aggregate principal amount of the Notes surrendered by such Noteholder relative to the aggregate principal amount of all Notes surrendered by all Noteholders pursuant to this Section 2.2(d)) of the New Senior Notes, the aggregate principal amount of which shall be equal to the aggregate principal amount of Notes surrendered by the Noteholders pursuant to this Section 2.2(d);
 - B. its respective Pro Rata Share (based on the aggregate principal amount of the Notes surrendered by such Noteholder relative to the aggregate principal amount of all Notes surrendered by all Noteholders pursuant to this Section 2.2(d)) of the New CBST Common Shares; and
 - C. if such 2025 Noteholder, 2026 Noteholder or New Senior Note-Electing Noteholder, as applicable, is an Early Supporting Noteholder, such Early Supporting Noteholder's Pro Rata Share

(based on the aggregate principal amount of the Notes surrendered by such Early Supporting Noteholder relative to the aggregate principal amount of all Notes surrendered by the Early Supporting Noteholders pursuant to this Section 2.2(d)) of: (A) the Early Consent Consideration, which shall be paid in accordance with Section 3.5; and (B) the right to receive the Additional Early Consent Consideration, which will be paid in accordance with Section 7.15(d) of the Amended and Restated Indenture, which, for greater certainty will be additional consideration for such Noteholder's Notes;

in full and final satisfaction of such Noteholder's Notes Claims, and CBST, CBST Canada 1 and each such Noteholder shall be deemed to have entered into the Standstill and Lock-up Agreement concurrently with the receipt of their New CBST Common Shares:

- (e) each New Convertible Note-Electing Noteholder shall: (i) be deemed to have surrendered its Notes in accordance with Section 3.2; and (ii) receive its respective Pro Rata Share (based on the aggregate principal amount of 2027 Notes surrendered by such New Convertible Note-Electing Noteholder relative to the aggregate principal amount of all 2027 Notes surrendered by New Convertible Note-Electing Noteholders pursuant to this Section 2.2(e)) of the New Convertible Notes issued by CBST and CBST Canada 1, the aggregate principal amount of which shall be equal to the principal amount of 2027 Notes surrendered by the New Convertible Note-Electing Noteholders pursuant to this Section 2.2(e), in full and final satisfaction of such Noteholder's Notes Claims;
- (f) the Notes and any related Notes Claims (including any make whole premiums, redemption premiums or similar premiums) shall be irrevocably and finally extinguished, discharged and settled and the Notes, the Existing Indenture and the other Notes Documents shall be deemed to be cancelled and terminated in all respects;
- (g) the releases referred to in Section 4.1 shall become effective; and
- (h) CBST Canada 1 and CBST Canada 2 will amalgamate under the CBCA (the "Amalgamation") and thereupon:
 - (i) the name of CBST Amalco will be "The Cannabist Company Holdings (Canada) Inc.";
 - (ii) the by-laws of CBST Amalco shall be the same as the by-laws of CBST Canada 2 immediately prior to the Effective Time;
 - (iii) the board of directors of CBST Amalco shall be the same as the directors of CBST Canada 2 immediately prior to the Effective Time; and
 - (iv) all shares of CBST Canada 1 shall be cancelled without any repayment of capital in respect thereof; no shares shall be issued by CBST Amalco in connection with such amalgamation; and all shares of CBST Canada 2

- prior to the Amalgamation shall be unaffected and shall continue as shares of CBST Amalco;
- (v) the stated capital account of the common shares of CBST Amalco shall be equal to the aggregate paid-up capital (for purposes of the Tax Act) of the common shares of CBST Canada 1 and CBST Canada 2 immediately prior to the Amalgamation;
- (vi) the provisions of subsection 186(a) to (g) of the CBCA shall apply to the Amalgamation with the result that:
 - A. the amalgamation of the amalgamating corporations and their continuance as one corporation becomes effective;
 - B. the property of each amalgamating corporation continues to be the property of CBST Amalco;
 - C. CBST Amalco continues to be liable for the obligations of each amalgamating corporation;
 - D. an existing cause of action, claim or liability to prosecution of either of the amalgamating corporations is unaffected;
 - E. a civil, criminal or administrative action or proceeding pending by or against an amalgamating corporation may be continued to be prosecuted by or against CBST Amalco;
 - F. a conviction against, or ruling, order or judgment in favour of or against, an amalgamating corporation may be enforced by or against CBST Amalco; and
 - G. the Articles of Arrangement are deemed to be the articles of incorporation of CBST Amalco and the Certificate of Arrangement is deemed to be the certificate of incorporation of CBST Amalco.

2.3 2027 Noteholder Election Mechanics

With respect to the surrender and exchange of 2027 Notes effected pursuant to Sections 2.2(d) and 2.2(e):

- (a) each 2027 Noteholder may elect to surrender and exchange its 2027 Notes for either:
 - (i) its Pro Rata Share of the New Senior Notes, New CBST Common Shares and, if applicable, the Early Consent Consideration and the right to receive the Additional Early Consent Consideration pursuant to Section 2.2(d) (such election being a "New Senior Note Election"); or
 - (ii) its Pro Rata Share of the New Convertible Notes pursuant to Section 2.2(e) (such election being a "New Convertible Note Election");

- (b) in order to make an election set forth in Section 2.3(a), each electing 2027 Noteholder must deposit with the Depositary, prior to the Election Deadline, a duly completed and executed Letter of Transmittal and Election Form indicating such 2027 Noteholder's election, which election shall be irrevocable and may not be withdrawn, together with any certificates representing the 2027 Notes held by such 2027 Noteholder and such additional documents and instruments as the Depositary, the Applicants or CBST may reasonably require; and
- (c) any 2027 Noteholder who does not make a valid New Senior Note Election or New Convertible Note Election prior to the Election Deadline in accordance with this Section 2.3, shall be deemed to have made a New Convertible Note Election.

2.4 Other Implementation Steps

- (a) The Applicants, CBST, and the other CBST Entities may undertake any other steps or transactions necessary or desirable to implement this Plan on the terms set out herein (as may be amended pursuant to the terms hereof) in any manner and on such date(s) and/or time(s) determined by the CBST Entities with the consent of the Requisite Supporting Noteholders, acting reasonably.
- (b) Without limiting the generality of Section 2.4(a), and subject in all cases to the consent of the Requisite Supporting Noteholders, to the extent that the Applicants, CBST and the other CBST Entities determine that it is not practicable to file and/or register, prior to the Effective Date, security documents and instruments that are required to be registered prior to the Effective Date pursuant to the Amended and Restated Indenture, such security documents and instruments may be filed or registered after the Effective Date, provided that, CBST, CBST Amalco and the other CBST Entities shall take best efforts to file and or register each such document and instrument as soon as practicable after the Effective Date and will not wait to register any such document or instrument until all such documents and instruments are available for registration; provided that, CBST, CBST Amalco and the other CBST Entities will use commercially reasonable efforts to provide any DACAs or any mortgages that are required pursuant to the Amended and Restated Indenture within 120 days from the Effective Date (or such longer time as agreed by the Requisite Supporting Noteholders).

2.5 Fractional Interests

- (a) The New Notes issued pursuant to this Plan shall be issued in minimum increments of U.S.\$1,000, and the amount of New Notes that each Noteholder shall be entitled to under this Plan shall in each case be rounded down to the nearest multiple of U.S.\$1,000 without compensation therefor.
- (b) No fractional New CBST Common Shares shall be issued pursuant to this Plan. In lieu of any fractional New CBST Common Shares, each Person otherwise entitled to a fractional interest in New CBST Common Shares will receive an aggregate amount of New CBST Common Shares rounded down to the nearest whole number increment.

2.6 Rounding of Cash

In any case where the aggregate cash amount payable to a particular Noteholder under the Arrangement would, but for this provision, include a fraction of a cent, the amount payable shall be rounded down to the nearest whole cent.

2.7 Calculations

All calculations made by the Applicants, CBST or Indenture Trustee pursuant to this Plan, absent manifest error, shall be conclusive, final and binding on all Persons affected by this Plan.

ARTICLE 3 EXCHANGE OF NOTES AND ISSUANCE OF NEW NOTES, NEW CBST COMMON SHARES, AND ANTI-DILUTIVE WARRANTS

3.1 Issuance of New Notes

The New Notes to be issued to the Noteholders pursuant to this Plan shall be made by way of issuance by CBST and CBST Canada 1 on the Effective Date of one or more global notes in respect of each of the New Senior Notes and New Convertible Notes, authenticated by the Indenture Trustee, and issued in the name of CDS (or its nominee) in respect of Noteholders, other than certain Holders of New Notes (or their Intermediaries) who, at the request of such Holders, will receive DRS Advices or definitive certificates representing the New Notes from the Indenture Trustee. Any definitive certificates shall be held by the Depositary until such time as the applicable Noteholder delivers a duly completed Letter of Transmittal and surrenders its Notes in accordance with Section 3.2.

3.2 Surrender of Notes

On the Effective Date, CDS (or its nominee) as a registered Noteholder on behalf of certain Noteholders shall surrender, or cause the surrender of, the Notes it holds in exchange for the portion of the consideration payable to it as a Noteholder pursuant to Section 2.2. At the time set out in Section 2.2, each Noteholder whose Notes are represented by a DRS Advice shall be deemed to have surrendered and transferred (free and clear of all liens and encumbrances), without any further action by the Noteholder, the Notes represented by DRS Advice to CBST and CBST Canada 1 in exchange for the portion of the consideration payable to it as a Noteholder pursuant to Section 2.2. On the Effective Date, each Noteholder holding a certificate representing Notes shall surrender, or cause the surrender of, the certificate(s) representing its respective Notes to the Depositary (with a duly completed Letter of Transmittal) in exchange for the portion of the consideration payable to it as a Noteholder pursuant to Section 2.2.

3.3 Issuance of New CBST Common Shares

- (a) The New CBST Common Shares to be issued pursuant to this Plan shall be deemed to be duly authorized, validly issued, fully paid and non-assessable. CBST will add to the stated capital account maintained in respect of the CBST Common Shares C\$1.00, in the aggregate, with respect to the issuance of the New CBST Common Shares.
- (b) Prior to the filing by the Applicants of the Articles of Arrangement, CBST shall deliver or cause to be delivered a treasury direction to the Transfer Agent, for the

benefit of the applicable Noteholders, that directs the Transfer Agent to issue the New CBST Common Shares to be distributed under this Plan and directs the Transfer Agent to cause the New CBST Common Shares to be distributed, subject to the provisions of the Standstill and Lock-up Agreement and Sections 3.3(c) and 3.7.

(c) The New CBST Common Shares to be issued under this Plan will be deposited with the Depositary and, subject to Section 3.2, delivered either (i) through the facilities of CDS who, in turn, will make delivery of the New CBST Common Shares to the ultimate beneficial recipients thereof pursuant to standing instructions and customary practices of CDS, as applicable, or (ii) by providing DRS Advices or confirmations in the name of the applicable recipient thereof (or its Intermediary) and registered in CBST's records which will be maintained by the Transfer Agent, following surrender of the Notes in accordance with Section 3.2 and Section 3.7.

3.4 Issuance of Anti-Dilutive Warrants

Prior to the filing by the Applicants of the Articles of Arrangement, CBST shall deliver or cause to be delivered: (a) a direction to the Warrant Agent, for the benefit of the CBST Shareholders, instructing the Warrant Agent to issue the Anti-Dilutive Warrants, to be distributed under this Plan; and (b) a reservation order to the Transfer Agent such that the number of CBST Common Shares issuable upon exercise of the Anti-Dilutive Warrants are reserved for issuance. Thereafter, the Warrant Agent shall distribute the Anti-Dilutive Warrants (x) through the facilities of CDS who, in turn, will make delivery of the Anti-Dilutive Warrants to the ultimate beneficial recipients thereof pursuant to standing instructions and customary practices of CDS, as applicable, and/or (v) by providing DRS Advices or confirmations in the name of the applicable recipient thereof (or its Intermediary) that is registered as a CBST Shareholder in CBST's records which are maintained by the Transfer Agent. The Anti-Dilutive Warrants will only be exercisable (i) outside the United States pursuant to Regulation S under the U.S. Securities Act, (ii) inside the United States, pursuant to exemptions from such registration requirements under the U.S. Securities Act and in compliance with any applicable U.S. state securities laws, or (iii) pursuant to an effective registration statement under the U.S. Securities Act, and in each case, in compliance with any applicable local securities Laws.

3.5 Payment of the Early Consent Consideration

Prior to the filing by the Applicants of the Articles of Arrangement, CBST shall deposit or cause to be deposited with the Depositary, for the benefit of the Early Supporting Noteholders, the Early Consent Consideration, and the Depository shall, subject to Section 3.2, deliver the Early Consent Consideration either (i) through the facilities of CDS who, in turn, will make delivery to the ultimate beneficial recipients thereof pursuant to standing instructions and customary practices of CDS, as applicable, and/or (ii) by delivering cheques, or in the case where the Early Consent Consideration is satisfied by the transfer of Verano Shares, certificates or DRS Advices representing such Verano Shares, in the name of the applicable recipient thereof (or its Intermediary) and registered in CBST's records which will be maintained by the Transfer Agent.

3.6 No Liability in Respect of Deliveries

(a) The Applicants, CBST or the other CBST Entities and their respective directors, officers, shareholders, agents or advisors, shall not have any liability or obligation in respect of any deliveries, directly or indirectly, from, as applicable, (i) the

Indenture Trustee, the Transfer Agent or the Warrant Agent, or (ii) the Intermediaries, in each case to the ultimate beneficial recipients of any consideration payable or deliverable by CBST or CBST Canada 1 pursuant to this Plan.

(b) The Indenture Trustee, the Depositary, CDS, the Transfer Agent or the Warrant Agent shall not incur, and each is hereby released from, any liability as a result of carrying out any provisions of this Plan and any actions related or incidental thereto, save and except for any gross negligence or wilful misconduct on its part (as determined by a final, non-appealable judgment of a court of competent jurisdiction). On the Effective Date, after the completion of the transactions set forth in Section 2.2, all duties and responsibilities of the Indenture Trustee arising under or related to the Notes shall be discharged except to the extent required in order to effectuate this Plan, including without limitation, as set forth in Section 3.1.

3.7 Deposit of Notes

- (a) At or before the Effective Time, each Noteholder may deposit all certificate(s) representing Notes with the Depositary to be held in escrow until the Arrangement takes effect. Upon surrender to the Depositary for cancellation of a certificate which immediately prior to the Effective Time represented Notes together with a duly completed and executed Letter of Transmittal, and such additional documents and instruments as the Depositary may reasonably require, the Noteholder of such surrendered certificate shall be entitled to receive in exchange therefor, and the Depositary shall deliver to such Noteholder, as soon as practicable after the Effective Time and in accordance with Sections 3.1 and 3.3(c), the number of New Notes and New CBST Common Shares to which it is entitled pursuant to this Plan, less any amounts withheld pursuant to Section 3.11, and any certificate representing Notes so surrendered shall forthwith be cancelled.
- (b) Until surrendered as contemplated by this Section 3.7, each certificate, which immediately prior to the Effective Time represented any Notes shall be deemed after the Effective Time to represent only the right to receive upon such surrender the aggregate principal amount of New Notes, number of New CBST Common Shares and amount of the Early Consent Consideration, as applicable, to which it is entitled pursuant to this Plan, less any amounts withheld pursuant to Section 3.11. Any such certificate formerly representing Notes not duly surrendered on or before the third anniversary of the Effective Date shall cease to represent a claim by or interest of any former Noteholder of any kind or nature against or in the Applicants, CBST or any other CBST Entity. On such anniversary date, all certificates representing the Notes shall be deemed to have been surrendered to CBST and any certificates for New Notes and New CBST Common Shares to which such former Holder was entitled, together with any entitlements accrued but unpaid interest thereon, shall be deemed to have been and shall be surrendered to CBST or CBST Canada 1 (or its successor) for cancellation, for no consideration.
- (c) Any certificate representing New Notes or New CBST Common Shares that has been returned to the Depositary or that otherwise remains unclaimed, in each case on or before the third anniversary of the Effective Date, shall cease to represent a right or claim of any kind or nature and the right of the Noteholder to receive the

consideration for the Notes pursuant to this Plan shall terminate and be deemed to be surrendered and forfeited to CBST and/or CBST Canada 1, as applicable, for no consideration.

3.8 Letter of Transmittal and Election Form

At the time of mailing the Circular or as soon as practicable thereafter, CBST shall forward the Letter of Transmittal and Election Form to registered Noteholders holding certificate(s) representing Notes in accordance with the Interim Order.

3.9 Lost Certificates

In the event any certificate which immediately prior to the Effective Time represented one or more Notes that were to be surrendered pursuant to Section 3.2, shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such certificate to be lost, stolen or destroyed, the Indenture Trustee may issue in exchange for such lost, stolen or destroyed certificate, a certificate or DRS Advice representing New Notes and the New CBST Common Shares deliverable in accordance with such holder's Letter of Transmittal. When authorizing such payment in exchange for any lost, stolen or destroyed certificate, the Person to whom any such certificate or DRS Advice is to be delivered shall as a condition precedent to the delivery of same, give a bond satisfactory to CBST, CBST Canada 1, the Indenture Trustee and the Transfer Agent in such sum as they may direct, acting reasonably, or otherwise indemnify CBST, CBST Canada 1, the Indenture Trustee and the Transfer Agent in a manner satisfactory to them against any claim that may be made against CBST, CBST Canada 1, the Indenture Trustee or the Transfer Agent with respect to the certificate alleged to have been lost, stolen or destroyed.

3.10 Securities Law Matters

- (a) The New Notes, New CBST Common Shares and New Guarantees issued in exchange for the Notes pursuant to this Plan shall be exempt from the registration requirements of the U.S. Securities Act pursuant to Section 3(a)(10) thereof;
- (b) The Anti-Dilutive Warrants will be issued to existing CBST Shareholders (excluding the recipients of New CBST Common Shares) in order to reduce the dilutive effect of the New CBST Common Shares. CBST will not receive any consideration for the issuance of the Anti-Dilutive Warrants. Therefore, the issuance of the Anti-Dilutive Warrants is not expected to constitute a "sale," as such term is defined in Section 2(a)(3) of the U.S. Securities Act and will not require registration under the U.S. Securities Act; and
- (c) The New Notes, New CBST Common Shares, and the Anti-Dilutive Warrants issued pursuant to this Plan shall be exempt from the prospectus requirements of Canadian Securities Laws pursuant to Section 2.11 of National Instrument 45-106 *Prospectus Exemptions* of the Canadian Securities Administrators.

3.11 Withholding Rights

a) CBST, the Applicants, the Indenture Trustee, the Transfer Agent, the Warrant Agent and the Intermediaries, and their respective agents, shall each be entitled to deduct and withhold from any consideration or other amount deliverable or

otherwise payable to any Person hereunder such amounts as they may be required to deduct or withhold with respect to such payment under the Tax Act, U.S. Code, or any provision of any applicable Tax Law, provided, that, any such right to deduct or withhold shall not otherwise change or modify their obligations in respect of withholding Taxes under the terms of the Existing Indenture or Amended and Restated Trust Indenture (or any supplements thereto) and any and all other documents. To the extent that Taxes or other amounts are so deducted or withheld and timely remitted to the Appropriate Governmental Entity, such deducted or withheld amounts shall be treated for all purposes hereof as having been paid to the relevant Person in respect of which such deduction and withholding was made. Such deducted or withheld amounts shall be remitted to the appropriate Governmental Entity.

- Notwithstanding the foregoing, if any of CBST, an Applicant, the Indenture Trustee, b) the Transfer Agent, the Intermediaries, their respective agents, or any other Person is required under the Tax Act, the U.S. Code, or any provision of any applicable Tax Law to deduct or withhold any Taxes in respect of any consideration or other amount deliverable or otherwise payable to any Noteholder (which definition shall, for purposes of this Section 3.11(b), include any beneficial holder of Notes) under this Plan ("Indemnified Taxes"), then CBST or an Applicant shall pay to such Noteholder such additional amounts in cash as necessary, so that after making or allowing for all required deductions and withholdings (including deductions and withholdings applicable to additional sums payable under this Section 3.11(b)) the Noteholder receives what it would have received (as an amount in cash, property or otherwise) had no such deductions or withholdings been required ("Additional Amounts"). Further, CBST, the Applicants and the Guarantors shall, without duplication of the foregoing, indemnify and hold harmless each Noteholder, within 10 days after written demand therefor, for the full amount of any Indemnified Taxes paid by the Noteholder (including with respect to any amounts paid under this Section 3.11(b)) and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Entity.
- c) Notwithstanding Section 3.11(b), no Additional Amounts or indemnity for Indemnified Taxes will be paid to Excluded Holders (as defined in the Amended and Restated Trust Indenture (excluding, for this purpose, the First Supplemental Indenture)).

ARTICLE 4 RELEASES

4.1 Release of Released Parties

As of the Effective Date, each of the Released Parties shall be released and discharged from all actions, causes of action, damages, judgments, executions, obligations, liabilities and Claims of any kind or nature whatsoever arising on or prior to the Effective Date in connection with the Notes, the Notes Claims, the Notes Documents, New CBST Common Shares, the Anti-

Dilutive Warrants, the Support Agreement, the Arrangement, the Plan, the CBCA Proceedings and any other proceedings commenced with respect to or in connection with the Plan, the transactions contemplated hereunder, and any other actions or matters related directly or indirectly to the foregoing (collectively, the "Released Claims"), provided that nothing in this paragraph shall release or discharge the following: (i) any of the Released Parties from or in respect of their respective obligations under the Plan, the New Notes (including the Amended and Restated Trust Indenture and the other New Notes Documents solely as they relate to the New Notes), the New Guarantees, the Warrant Agency Agreement, the New CBST Common Shares, the Anti-Dilutive Warrants, or any Order or document ancillary to any of the foregoing, (ii) any director or officer of CBST, the Applicants or any other CBST Entity of their right to indemnity, insurance claims, and employment related rights or claims, (iii) any inter-company obligations between or among CBST, the Applicants or any other CBST Entity on or before the Effective Date; or (iv) any act or omission arising out of any Released Party's gross negligence, actual and intentional fraud, willful misconduct, or criminal acts (as determined by a final non-appealable order from a court of competent jurisdiction) (collectively, the "Release Carve-Outs"). The foregoing release shall not be construed to prohibit a party in interest from seeking to enforce the terms of this Plan or any contract or agreement entered into pursuant to, in connection with or contemplated by this Plan.

4.2 Injunctions

All Persons are permanently and forever barred, estopped, stayed and enjoined, on and after the Effective Date, with respect to any and all Released Claims, from: (i) commencing, conducting or continuing in any manner, directly or indirectly, any action, suits, demands or other proceedings of any nature or kind whatsoever of any Person against the Released Parties, as applicable; (ii) enforcing, levying, attaching, collecting or otherwise recovering or enforcing by any manner or means, directly or indirectly, any judgment, award, decree or order against the Released Parties; (iii) creating, perfecting, asserting or otherwise enforcing, directly or indirectly, any lien or encumbrance of any kind against the Released Parties or their property; or (iv) taking any actions to interfere with the implementation or consummation of this Plan or the transactions contemplated hereunder; provided, however, that the foregoing shall not apply to the enforcement of any obligations under this Plan or any document, instrument or agreement executed to implement this Plan or issue the New Notes, the New Guarantees, the New CBST Common Shares, and/or the Anti-Dilutive Warrants.

ARTICLE 5 CONDITIONS PRECEDENT AND IMPLEMENTATION

5.1 Conditions to Plan Implementation

The implementation of this Plan and the occurrence of the Effective Date shall be conditional upon the fulfilment, satisfaction or waiver (to the extent permitted by Section 5.2) of the following conditions:

- (a) the Court shall have granted the Final Order, the implementation, operation or effect of which shall not have been stayed or vacated;
- (b) the Support Agreement shall not have been validly terminated in accordance with Section 9 of the Support Agreement; and

(c) all conditions to implementation contained in Section 8 of the Support Agreement shall have been satisfied or waived in accordance with the terms of the Support Agreement, other than the condition contemplated by Section 8(1)(a) of the Support Agreement.

5.2 Waiver of Conditions

CBST, the Applicants and the Requisite Supporting Noteholders may at any time and from time to time jointly waive the fulfilment or satisfaction, in whole or in part, of the conditions set out in Section 5.1 to the extent and on such terms as they may agree, provided however that the condition set out in Section 5.1(a) cannot be waived.

5.3 Effectiveness

This Plan shall become effective in the sequence described in Section 2.2 on the filing of the Articles of Arrangement and the issuance of the Certificate of Arrangement and shall, from and after the Effective Time, be binding on and enure to the benefit of the Persons described in Section 1.7, the Released Parties, and all other Persons named or referred to in, or subject to, this Plan and their respective successors and assigns and their respective heirs, executors, administrators and other legal representatives, and their respective successors and assigns. The Articles of Arrangement shall be filed no earlier than five (5) Business Days following the Court granting the Final Order unless agreed otherwise by the Applicants, CBST and the Requisite Supporting Noteholders. The Articles of Arrangement shall be filed and the Certificate of Arrangement shall be issued in each case with respect to the Arrangement in its entirety. The Certificate of Arrangement shall be conclusive evidence that the conditions in Section 5.1 have been satisfied or waived, the Arrangement has become effective and that each of the provisions in Section 2.2 have become effective in the sequence set forth therein. No portion of this Plan shall take effect with respect to any party or Person until the Effective Time.

5.4 Effect of Non-Occurrence of Effective Date

If the Effective Date does not occur on or before the valid termination of the Support Agreement in accordance with its terms, unless the Applicants, CBST and the Requisite Supporting Noteholders agree in writing, then (a) this Plan shall be null and void in all respects; (b) any settlement or compromise embodied in this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null and void; and (c) the obligations of CBST and CBST Canada 1 and the Existing Guarantors with respect to the Notes, the Notes Documents and the Notes Claims shall remain unchanged and nothing contained in this Plan shall constitute or be deemed to constitute a waiver or release of any Notes Claims or otherwise.

ARTICLE 6 GENERAL

6.1 Deemed Consents, Waivers and Agreements

- (a) At the Effective Time, each Noteholder shall be deemed to have consented and agreed to all of the provisions of this Plan in its entirety; and
- (b) Each CBST Entity, Noteholder and beneficial owner of Notes shall be deemed to have executed and delivered to the other parties all consents, releases,

assignments and waivers, statutory or otherwise, required to implement and carry out this Plan in its entirety.

6.2 Waiver of Defaults

From and after the Effective Time, all Persons named or referred to in, or subject to, this Plan shall be deemed to have consented and agreed to all of the provisions of this Plan in its entirety. Without limiting the foregoing, from and after the Effective Time, all Persons shall be deemed to have:

- waived any and all defaults or events of default, third-party change of control rights (a) or any non-compliance with any covenant, warranty, representation, term, provision, condition or obligation, expressed or implied, in any contract, instrument, credit document, lease, licence, guarantee, seller note, vendor takeback note, security agreement, subordination agreement, agreement for sale or other agreement, written or oral, in each case relating to, arising out of, or in connection with, the Notes, the Notes Documents, this Plan, the Arrangement, the Support Agreement, the transactions contemplated thereby or hereunder, the CBCA Proceedings and any other proceedings commenced with respect to or in connection with this Plan. Any and all notices of default and demands for payment or any step or proceeding taken or commenced in connection with any of the foregoing shall be deemed to have been rescinded, irrevocably waived and of no further force or effect; provided that nothing shall be deemed to excuse the Applicants, CBST or the other CBST Entities, and their respective successors and assigns, from performing their obligations under this Plan, the New Notes Documents or any other contract or agreement entered into pursuant to, in connection with, or contemplated by, this Plan; and
- (b) agreed that if there is any conflict between the provisions of any agreement or other arrangement, written or oral, existing between such Person and the Applicants, CBST or the other CBST Entities prior to the Effective Date and the provisions of this Plan, then the provisions of this Plan take precedence and priority and the provisions of such agreement or other arrangement are deemed to be amended accordingly.

For greater certainty, nothing in this Plan shall release or waive Claims in respect of the Release Carve-Outs.

6.3 Compliance with Deadlines

The Applicants have the right to waive strict compliance with any election or other deadlines pursuant to this Plan (with the prior consent of the Requisite Supporting Noteholders, which shall not be unreasonably withheld or delayed), and shall be entitled to waive any deficiencies with respect to any forms or other documentation submitted pursuant to this Plan.

6.4 Paramountcy

From and after the Effective Date, any conflict between this Plan and the covenants, warranties, representations, terms, conditions, provisions or obligations, expressed or implied, of any contract, mortgage, security agreement, indenture, trust indenture, guarantee, subordination agreement, loan agreement, commitment letter, by-laws or other agreement, written or oral

existing between one or more of the Noteholders and any one or more of the Applicants, CBST and/or the Existing Guarantors with respect to the Notes Documents as at the Effective Date shall be deemed to be governed by the terms, conditions and provisions of this Plan and the Final Order, which shall take precedence and priority. In the event of any conflict between this Plan and the New Notes Documents, the New Notes Documents shall take precedence and priority.

6.5 Deeming Provisions

In this Plan, the deeming provisions are not rebuttable and are conclusive and irrevocable.

6.6 Amendment or Modification of Plan

- (a) The Applicants reserve the right to amend, restate, modify and/or supplement this Plan at any time and from time to time, in a manner consistent with the Support Agreement or otherwise acceptable to the Requisite Supporting Noteholders; provided that any such amendment, restatement, modification or supplement must be contained in a written document that is (i) filed with the Court and, subject to Section 6.6(c) below, if made following the Noteholders Meeting, approved by the Court, and (ii) communicated to the Noteholders in the manner required by the Court (if so required).
- (b) Any amendment, restatement, modification or supplement to this Plan may be proposed by the Applicants, in a manner consistent with the Support Agreement or otherwise acceptable to the Requisite Supporting Noteholders, at any time prior to or at the Noteholders Meeting, with or without any prior notice or communication (other than as may be required under the Interim Order), and if so proposed and accepted at the Noteholders Meeting, shall become part of this Plan for all purposes.
- (c) Any amendment, restatement, modification or supplement to this Plan may be made following the Noteholders Meeting by the Applicants, in a manner consistent with the Support Agreement or otherwise acceptable to the Requisite Supporting Noteholders, without requiring filing with, or approval of, the Court; provided that it concerns a matter which is of an administrative nature and is required to better give effect to the implementation of this Plan and is not materially adverse to the financial or economic interests of any of the Noteholders.
- (d) Any amendment, restatement, modification or supplement to this Plan that is approved or directed by the Court following the Noteholders Meeting shall be effective only if it is consented to by each of the Applicants, CBST and the Requisite Supporting Noteholders.
- (e) Any amendment, supplement or modification of this Plan that materially adversely affects any of the Supporting Noteholders in a disproportionate manner shall require the written approval of the adversely affected Supporting Noteholder.

6.7 Notices

Any notice or other communication to be delivered hereunder must be in writing and refer to this Plan and may, as hereinafter provided, be made or given by personal delivery, prepaid mail or email addressed to the respective parties as follows:

(a) if to the Applicants or CBST, at:

321 Billerica Road, Suite 204 Chelmsford, MA 01824

Attention: David Hart / David Sirolly

Email: david.hart@cannabistcompany.com

david.sirolly@cannabistcompany.com

with a copy (which shall not be deemed notice) to:

Stikeman Elliott LLP 5300 Commerce Court West, 199 Bay Street Toronto, Ontario M5L 1B9

Attention: Martin Langlois / Lee Nicholson / Shawn Blundell

Email: mlanglois@stikeman.com /

<u>leenicholson@stikeman.com /</u> sblundell@stikeman.com

and

Dorsey & Whitney LLP 66 Wellington Street West, Suite 3400 Toronto, Ontario M5K 1E6

Attention: Erin McCrady / James Guttman / Robert Van de Mark

<u>Email</u>: mccrady.erin@dorsey.com /

guttman.james@dorsey.com / vandemark.bob@dorsey.com

(b) if to any Supporting Noteholder, at:

Goodmans LLP
Bay Adelaide Centre - West Tower
333 Bay Street, Suite 3400
Toronto, ON M5H 2S7

Attention: Brendan O'Neill and Bradley Wiffen

<u>Email</u>: boneill@goodmans.ca / bwiffen@goodmans.ca

and

Feuerstein Kulick LLP 420 Lexington Avenue, Suite 204 New York, New York 10170

Attention: Samantha Gleit / Anan Kahari

Email: samantha@dfmklaw.com / <u>akahari@dfmklaw.com</u>

(c) if to the Indenture Trustee, the Transfer Agent, the Warrant Agent or the Depositary, at:

Odyssey Trust Company Stock Exchange Tower 1230 – 300 5th Avenue SW Calgary AB T2P 3C4

Attn: Corporate Trust

or to such other address as any party above may from time to time notify the others in accordance with this Section 6.7. In the event of any strike, lock-out or other event which interrupts postal service in any part of Canada or the United States, all notices and communications during such interruption may only be given or made by personal delivery, courier or by email and any notice or other communication given or made by prepaid mail within the five (5) Business Day period immediately preceding the commencement of such interruption, unless actually received, shall be deemed not to have been given or made. Any such notices and communications so given or made, in the case of notice by way of personal delivery, courier or email, shall be deemed to have been given or made and to have been received on the day of delivery, couriering or of emailing, as applicable, if received on a Business Day before 5:00 p.m. (local time), or on the next following Business Day if received after 5:00 p.m. (local time) on a Business Day or at any time on a non-Business Day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the fifth Business Day following the date on which such notice or other communication is mailed or couriered. The unintentional failure by the Applicants to give a notice contemplated hereunder to any particular Noteholder shall not invalidate this Plan or any action taken by any Person pursuant to this Plan.

6.8 Consent of Supporting Noteholders

For the purposes of this Plan, any matter requiring the agreement, waiver, consent or approval of the Supporting Noteholders or Requisite Supporting Noteholders shall be deemed to have been agreed to, waived, consented to or approved by the Supporting Noteholders or Requisite Supporting Noteholders, as applicable, if such matter is agreed to, waived, consented to or approved in writing (which can be by way e-mail) by any of the Supporting Noteholders Advisors on behalf of the Supporting Noteholders or the Requisite Supporting Noteholders, as applicable, provided that such Supporting Noteholders Advisor confirms in writing (which can be by way of e-mail) that it is providing such agreement, consent, waiver or approval on behalf of the Supporting Noteholders or the Requisite Supporting Noteholders, as applicable.

6.9 Further Assurances

Notwithstanding that the transactions and events set out herein will occur and be deemed to occur in the order set out in this Plan without any further act or formality (except as provided herein), each of the Persons named or referred to in, affected by or subject to, this Plan will make, do and execute, or cause to be made, done and executed, all such further acts, deeds, agreements, transfers, assurances, instruments or documents as may reasonably be required by the Applicants, CBST or the Requisite Supporting Noteholders to carry out the full intent and meaning of this Plan and to give effect to the transactions contemplated herein.

IN THE MATTER OF AN APPLICATION UNDER SECTION 192 OF THE CANADA BUSINESS CORPORATIONS ACT, R.S.C. 1985, c. C-44, AS AMENDED

AND IN THE MATTER OF RULE 14.05(2) OF THE RULES OF CIVIL PROCEDURE

AND IN THE MATTER OF A PROPOSED PLAN OF ARRANGEMENT OF THE CANNABIST COMPANY HOLDINGS (CANADA) INC. AND 16834434 CANADA INC., et al.

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

Proceeding commenced at Toronto

FINAL ORDER

STIKEMAN ELLIOTT LLP

Barristers & Solicitors 5300 Commerce Court West 199 Bay Street Toronto, Canada M5L 1B9

Lee Nicholson LSO#: 66412I Email: leenicholson@stikeman.com

Tel: +1 416-869-5230

Eliot Kolers LSO#: 38304R Email: ekolers@stikeman.com

Tel: 416-869-5637

Philip Yang LSO#: 820840 Email: PYang@stikeman.com

Tel: +1 416-869-5593

Brittney Ketwaroo LSO#: 89781K Email: bketwaroo@stikeman.com

Tel: +1 416-869-5524

Lawyers for the Applicants

SCHEDULE "B"

Statutory Authorities

Companies' Creditors Arrangement Act, RSC 1985, c C-36

General power of court

11 Despite anything in the <u>Bankruptcy and Insolvency Act</u> or the <u>Winding-up and Restructuring Act</u>, if an application is made under this Act in respect of a debtor company, the court, on the application of any person interested in the matter, may, subject to the restrictions set out in this Act, on notice to any other person or without notice as it may see fit, make any order that it considers appropriate in the circumstances.

Stays, etc. — other than initial application

11.02(2) A court may, on an application in respect of a debtor company other than an initial application, make an order, on any terms that it may impose,

- (a) staying, until otherwise ordered by the court, for any period that the court considers necessary, all proceedings taken or that might be taken in respect of the company under an Act referred to in paragraph (1)(a);
- (b) restraining, until otherwise ordered by the court, further proceedings in any action, suit or proceeding against the company; and
- (c) prohibiting, until otherwise ordered by the court, the commencement of any action, suit or proceeding against the company.

Burden of proof on application

- **11.02(3)** The court shall not make the order unless
 - (d) the applicant satisfies the court that circumstances exist that make the order appropriate; and
 - (e) in the case of an order under subsection (2), the applicant also satisfies the court that the applicant has acted, and is acting, in good faith and with due diligence.

Interim financing

11.2 (1) On application by a debtor company and on notice to the secured creditors who are likely to be affected by the security or charge, a court may make an order declaring that all or part of the company's property is subject to a security or charge — in an amount that the court considers appropriate — in favour of a person specified in the order who agrees to lend to the company an amount approved by the court as being required by the company, having regard to its cash-flow statement. The security or charge may not secure an obligation that exists before the order is made.

Factors to be considered

- (4) In deciding whether to make an order, the court is to consider, among other things,
 - (f) the period during which the company is expected to be subject to proceedings under this Act;
 - (g) how the company's business and financial affairs are to be managed during the proceedings;
 - (h) whether the company's management has the confidence of its major creditors;
 - (i) whether the loan would enhance the prospects of a viable compromise or arrangement being made in respect of the company;
 - (j) the nature and value of the company's property;
 - (k) whether any creditor would be materially prejudiced as a result of the security or charge; and
 - (I) the monitor's report referred to in paragraph 23(1)(b), if any.

Restriction on disposition of business assets

36 (1) A debtor company in respect of which an order has been made under this Act may not sell or otherwise dispose of assets outside the ordinary course of business unless authorized to do so by a court. Despite any requirement for shareholder approval, including one under federal or provincial law, the court may authorize the sale or disposition even if shareholder approval was not obtained.

Notice to creditors

(2) A company that applies to the court for an authorization is to give notice of the application to the secured creditors who are likely to be affected by the proposed sale or disposition.

Factors to be considered

- (3) In deciding whether to grant the authorization, the court is to consider, among other things,
 - (a) whether the process leading to the proposed sale or disposition was reasonable in the circumstances;
 - (b) whether the monitor approved the process leading to the proposed sale or disposition;
 - (c) whether the monitor filed with the court a report stating that in their opinion the sale or disposition would be more beneficial to the creditors than a sale or disposition under a bankruptcy;
 - (d) the extent to which the creditors were consulted:

- (e) the effects of the proposed sale or disposition on the creditors and other interested parties; and
- (f) whether the consideration to be received for the assets is reasonable and fair, taking into account their market value.

Additional factors — related persons

- (4) If the proposed sale or disposition is to a person who is related to the company, the court may, after considering the factors referred to in subsection (3), grant the authorization only if it is satisfied that
 - (a) good faith efforts were made to sell or otherwise dispose of the assets to persons who are not related to the company; and
 - (b) the consideration to be received is superior to the consideration that would be received under any other offer made in accordance with the process leading to the proposed sale or disposition.

Related persons

- (5) For the purpose of subsection (4), a person who is related to the company includes
 - (a) a director or officer of the company;
 - (b) a person who has or has had, directly or indirectly, control in fact of the company; and
 - (c) a person who is related to a person described in paragraph (a) or (b).

Assets may be disposed of free and clear

(6) The court may authorize a sale or disposition free and clear of any security, charge or other restriction and, if it does, it shall also order that other assets of the company or the proceeds of the sale or disposition be subject to a security, charge or other restriction in favour of the creditor whose security, charge or other restriction is to be affected by the order.

Restriction — employers

(7) The court may grant the authorization only if the court is satisfied that the company can and will make the payments that would have been required under <u>paragraphs 6(5)(a)</u> and <u>(6)(a)</u> if the court had sanctioned the compromise or arrangement.

Restriction — intellectual property

(8) If, on the day on which an order is made under this Act in respect of the company, the company is a party to an agreement that grants to another party a right to use intellectual property that is included in a sale or disposition authorized under subsection (6), that sale or disposition does not affect that other party's right to use the intellectual property — including the other party's right to enforce an exclusive use — during the term of the agreement, including any period for which the other party extends the agreement as of right, as long as the other party continues to perform its obligations under the agreement in relation to the use of the intellectual property.

Courts of Justice Act, RSO 1990, c C.43

Declaratory orders

97 The Court of Appeal and the Superior Court of Justice, exclusive of the Small Claims Court, may make binding declarations of right, whether or not any consequential relief is or could be claimed.

Sealing documents

137(2) A court may order that any document filed in a civil proceeding before it be treated as confidential, sealed and not form part of the public record

IN THE MATTER OF THE COMPANIES' CREDITORS ARRANGEMENT ACT, R.S.C. 1985, c.C-36 AS AMENDED

AND IN THE MATTER OF A PLAN OF COMPROMISE OR ARRANGEMENT OF SHAW-ALMEX INDUSTRIES LIMITED AND SHAW ALMEX FUSION, LLC

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

Proceedings commenced at Toronto

FACTUM OF THE APPLICANTS

RECONSTRUCT LLP

80 Richmond Street Wes Suite 1700 Toronto, ON M5H 2A4

Brendan Bissell LSO No. 40354V bbissell@reconllp.com
Tel: 416.613.0066

Simran Joshi LSO No. 89775A sjoshi@reconllp.com Tel: 416.613.6589

Lawyers for the Applicants